# Variable Selection and Inference for Multi-period Forecasting Problems

## M. HASHEM PESARAN ANDREAS PICK Allan Timmermann

## CESIFO WORKING PAPER NO. 2543 CATEGORY 10: EMPIRICAL AND THEORETICAL METHODS FEBRUARY 2009

An electronic version of the paper may be downloaded • from the SSRN website: www.SSRN.com • from the RePEc website: www.RePEc.org • from the CESifo website: www.CESifo-group.org/wp

## Variable Selection and Inference for Multi-period Forecasting Problems

## Abstract

This paper conducts a broad-based comparison of iterated and direct multi-step forecasting approaches applied to both univariate and multivariate models. Theoretical results and Monte Carlo simulations suggest that iterated forecasts dominate direct forecasts when estimation error is a first-order concern, i.e. in small samples and for long forecast horizons. Conversely, direct forecasts may dominate in the presence of dynamic model misspecification. Empirical analysis of the set of 170 variables studied by Marcellino, Stock and Watson (2006) shows that multivariate information, introduced through a parsimonious factor-augmented vector autoregression approach, improves forecasting performance for many variables, particularly at short horizons.

| M. Hashem Pesaran    | Andreas Pick              |
|----------------------|---------------------------|
| Faculty of Economics | De Nederlandsche Bank and |
| Cambridge University | Faculty of Economics      |
| Sidgwick Avenue      | Cambridge University      |
| Cambridge, CB3 9DD   | Sidgwick Avenue           |
| United Kingdom       | Uk - Cambridge, CB3 9DD   |
| mhp1@econ.cam.ac.uk  | ap236@econ.cam.ac.uk      |

Allan Timmermann University of California San Diego Rady School of Management Otterson Hall, Room 4S146 9500 Gilman Dr. #0553 USA - La Jolla, CA 92093-0553 atimmerm@ucsd.edu

January 26, 2009

We thank Alessio Sancetta and seminar participants at the 2008 Rio Forecasting Conference at Fundacao Getulio Vargas for comments and suggestions on the paper. Timmermann acknowledges support from CREATES, funded by the Danish National Research Foundation. This paper was written while the second author was Sinopia Research Fellow at the University of Cambridge. He acknowledges financial support from Sinopia, quantitative specialist of HSBC Global Asset Management. The opinions expressed in this paper do not necessarily reflect those of DNB.

### 1 Introduction

Economists are commonly asked to forecast uncertain outcomes multiple periods ahead in time. For example, in a recession state a policy maker may want to know when the economy will recover and so is interested in forecasts of output growth for, say, horizons of 1, 3, 6, 12, and 24 months. Similarly, fixed-income investors are interested in comparing forecasts of spot rates multiple periods ahead against current long-term interest rates in order to arrive at an optimal investment strategy.

Given the importance of the horizon to many forecasting problems, it is not surprising that a substantial literature has considered the multi-step forecasting problem, see Cox (1961), Brown and Mariano (1989), Clements and Hendry (1998), Findley (1983), Schorfheide (2005), and Ullah (2004), with Bhansali (1999) providing a survey, and Ing (2003) giving asymptotic results for stationary ARs.

Two basic strategies exist for generating multi-period forecasts. The first approach is to estimate a dynamic model for data observed at the highest available frequency, e.g. monthly, and then use the chain rule to generate a forecast at the desired horizon,  $h \ge 1$ . Under this "iterated" or "indirect" approach, the forecasting specification is the same across all forecast horizons; only the number of iterations changes with h. In the context of univariate specifications, typically autoregressive models are used to this end, while if multivariate specifications are considered, vector autoregressions must be used. The second approach is to estimate a model for the variable measured h-periods ahead as a function of current information. This leads to so-called "direct" forecasts. Under this approach, the forecasting model and its estimates will typically vary across different forecast horizons.

Each approach has advantages and drawbacks. For a given model the iterated approach leads to more efficient estimates since it makes use of data recorded at the highest available frequency and so uses the largest available sample size. Conversely, if the model is misspecified, due, for example, to using an incorrect lag order, iterating the model multiple steps ahead can attenuate any existing biases. Direct forecasts are less efficient, but also more likely to be robust to model misspecification as they are typically linear projections of current realizations on past data. Direct forecasts introduce new problems, however, due to the overlap in data resulting from h > 1, which affect the covariance of the forecast errors.

Given such trade-offs, which approach delivers the best forecasts, the direct or the iterated, is therefore ultimately an empirical question. For univariate forecasting problems, as pointed out by Marcellino, Stock and Watson (2006) the performance of the two methods will depend on the sample size, forecast horizon and method used to select lag lengths for the forecasting models. In a multivariate setting it also becomes important how the potentially high-dimensional variable selection search is conducted and how multi-step forecasts of additional predictor variables are generated.

For multivariate specifications of even modest dimension, a global model specification search very rapidly becomes intractable unless the problem is further constrained: with d potential regressors, there are  $2^d$  different linear models and with d easily in the hundreds it is infeasible to evaluate every possible model.

To deal with this dimensionality problem, we propose a factor-augmented VAR approach to iterated forecasting that builds on the work by Bernanke, Boivin and Eliasz (2005) and Stock and Watson (2005). This limits the model specification search to consider inclusion of only a few common factors extracted from different categories of economic variables such as income/output, employment, construction/inventories, interest rates/asset prices and nominal prices/wages. This means that in addition to past values of the predicted variable itself relatively few potential predictors need to be considered.

How close the best forecasting models approximate the "true" data generating process also plays a role since model misspecification can change the relative performance of the iterated and direct forecasts. Finally, we show that how the direct forecasts account for data overlap can affect the results. We propose modifications to existing model selection methods that account for such data overlap.

Empirically, we confirm the finding reported by Marcellino, Stock and Watson (2006) that the iterated forecasts are best overall among the univariate forecasting methods. However, we also find that forecasts generated by the factor-augmented VARs generally perform better than the univariate forecasts, the important exception being variables tracking prices and wages. This suggests that it is helpful to extend the forecasting models beyond purely univariate schemes and include multivariate information.

The main contributions of the paper are as follows: First, for simple dynamic models such as the AR(1) specification we present a new theoretical result establishing conditions under which both the iterated and the direct forecasts are unconditionally unbiased. A key condition turns out to be symmetry of the innovations. Second, we extend existing model selection criteria such as the AIC and the BIC to account for the overlap introduced by the direct forecasting method which affects the covariance of the forecast errors and so can lead to different models being chosen in small samples. Third, we undertake a comprehensive study of model selection methods and shed light, both through Monte Carlo simulations and through empirical results, on which approach works best in the recursive modeling of a variety of economic variables. We consider univariate models as well as multivariate factor augmented vector autoregressive specifications hitherto not used in the context of direct and indirect multi-period forecast comparisons.

The outline of the paper is as follows. Section 2 discusses the theoretical trade-offs associated with direct versus iterated forecasts. Section 3 sets up the forecasting problem for the univariate and multivariate case, while Sec-

tion 4 deals with the model selection issue. Section 5 presents some Monte Carlo results, while Section 6 describes our empirical findings using the data originally analyzed by Marcellino, Stock and Watson (2006). Section 7 concludes. Technical results are provided in appendices.

## 2 Comparison of Direct and Iterated Forecasts for the AR(1) Model

In this section we introduce the iterated and direct forecasting approaches in the context of the simplest possible dynamic model, namely an AR(1) specification. Next, we establish a new result showing that both approaches lead to forecasts that are unconditionally unbiased provided that the innovations to the underlying data generating process are symmetrically distributed. Finally, we characterize the relative efficiency of the two approaches in finite samples.

The analysis is conducted under the assumption that a stationary AR(1) model of the following form is the correct specification:

$$y_t = a + \phi y_{t-1} + \varepsilon_t, \quad |\phi| < 1, \ \varepsilon_t \sim iid(0, \sigma_{\varepsilon}^2).$$
 (1)

Letting  $\mu = a/(1-\phi)$ , we can use the equivalent representation

$$y_t = \mu + u_t, \text{ where}$$
(2)  
$$u_t = \sum_{i=0}^{\infty} b_i \varepsilon_{t-i}, \quad b_i = \phi^i.$$

Alternatively, for purposes of the direct forecasting approach that projects  $y_t$  on  $y_{t-h}$ , we can write

$$y_{t} = a\left(\frac{1-\phi^{h}}{1-\phi}\right) + \phi^{h}y_{t-h} + v_{t}, \qquad (3)$$
$$\equiv a_{h} + \phi_{h}y_{t-h} + v_{t}, \quad \text{where}$$
$$v_{t} = \sum_{j=0}^{h-1} \phi^{j}\varepsilon_{t-j}.$$

Notice that  $v_t$  will follow an MA(h-1) process even when  $\varepsilon_t$  is serially uncorrelated due to the data overlap resulting from h > 1.

#### 2.1 Iterated and Direct Approach

Suppose that we are interested in forecasting  $y_{T+h}$  based on information up to period T and that the T+h observations  $y_t$ ,  $t = -h+1, -h+2, \ldots, T$  are available for model estimation. Denote the h-step ahead forecast of  $y_{T+h}$  by

 $\hat{y}_{T+h}$  if the iterated procedure is used, and by  $\tilde{y}_{T+h}$  if the direct method is employed.

Under the iterated procedure

$$\hat{y}_{T+h} = \hat{a}_T \left( \frac{1 - \hat{\phi}_T^h}{1 - \hat{\phi}_T} \right) + \hat{\phi}_T^h y_T, \tag{4}$$

where  $\hat{a}_T$  and  $\hat{\phi}_T$  are the estimators of a and  $\phi$  obtained from the OLS regression (1) of  $y_t$  on an intercept and  $y_{t-1}$  using the observations  $y_t$ ,  $t = -h+1, -h+2, \ldots, T$ .

The direct forecast of  $y_{T+h}$  is given by

$$\tilde{y}_{T+h} = \tilde{a}_{h,T} + \phi_{h,T} y_T, \tag{5}$$

where  $\tilde{a}_{h,T}$  and  $\phi_{h,T}$  are the OLS estimators of  $a_h$  and  $\phi_h$  obtained from the regression (3) of  $y_t$  on an intercept and  $y_{t-h}$  using the same sample observations,  $y_t$ ,  $t = -h + 1, -h + 2, \ldots, T$ .

Specifically, under the iterated method,

$$(T+h-1)^{-1}\sum_{t=-h+2}^{T} y_t = \hat{a}_T + \hat{\phi}_T (T+h-1)^{-1} \sum_{t=-h+2}^{T} y_{t-1},$$

or, equivalently,

$$\bar{y}_{h:T} = \hat{a}_T + \phi_T \ \bar{y}_{h:T,-1},$$
(6)

where

$$\hat{\phi}_T = \frac{\sum_{t=-h+2}^T y_t (y_{t-1} - \bar{y}_{h:T,-1})}{\sum_{t=-h+2}^T (y_{t-1} - \bar{y}_{h:T,-1})^2}.$$

Similarly, for the direct method we have

$$T^{-1} \sum_{t=1}^{T} y_t = \tilde{a}_{h,T} + \tilde{\phi}_{h,T} T^{-1} \sum_{t=1}^{T} y_{t-h}$$

or, equivalently (noting that  $\bar{y}_T = \bar{y}_{1:T}$ ),

$$\bar{y}_T = \tilde{a}_{h,T} + \tilde{\phi}_{h,T} \bar{y}_{T,-h},\tag{7}$$

where

$$\tilde{\phi}_{h,T} = \frac{\sum_{t=1}^{T} y_t (y_{t-h} - \bar{y}_{T,-h})}{\sum_{t=1}^{T} (y_{t-h} - \bar{y}_{T,-h})^2}.$$
(8)

We next establish conditions under which both the direct and indirect forecasts are unconditionally unbiased:

**Proposition 1** Suppose data is generated by the stationary AR(1) process, (1) and define the h-step ahead forecast errors from the iterated and direct methods,

$$\hat{e}_{T+h} = y_{T+h} - \hat{y}_{T+h},$$

and

$$\tilde{e}_{T+h} = y_{T+h} - \tilde{y}_{T+h}$$

where the iterated h-step forecast,  $\hat{y}_{T+h}$ , and the direct forecast,  $\tilde{y}_{T+h}$ , are given, for example, by (4) and (5), respectively. Assume that  $u_t$  and  $v_t$ , defined in (2) and (3), are symmetrically distributed around zero, have finite second-order moments and expectations of  $\hat{\phi}_T$  and  $\tilde{\phi}_{h,T}$  exist. Then for any finite T and h we have

$$\mathcal{E}(\hat{e}_{T+h}) = \mathcal{E}(\tilde{e}_{T+h}) = 0.$$

A proof of this proposition is provided in Appendix A.

The result is surprising since neither set of estimators  $(\hat{a}_T, \hat{\phi}_T)$  and  $(\tilde{a}_{h,T}, \tilde{\phi}_{h,T})$  is unbiased. However, the main intuition behind the results follows from the fact that, under the stated assumptions,

$$\mathcal{E}(\hat{\mu}_T - \mu) = 0,$$

and

$$\operatorname{E}\left(\frac{\hat{a}_T}{1-\hat{\phi}_T}\right) = \frac{a}{1-\phi}.$$

The proposition generalizes the known result in the literature for h = 1 established, for example, by Fuller (1996) to multi-step ahead forecasts. For h = 1, Pesaran and Timmermann (2005) also show that forecast errors are unconditionally unbiased for symmetrically distributed error processes even in the presence of breaks in the autocorrelation coefficient,  $\phi$ , so long as  $\mu$  is stable over the estimation sample.

Notice that the result holds only unconditionally and that the direct and iterated forecasts are not typically conditionally unbiased for a given value of  $y_T \neq \mu$ , as noted by Phillips (1987).

In comparing the iterated and direct forecasts it is worth noting that when  $\phi$  is positive and not too close to unity, for even moderately large values of h,  $\hat{\phi}_T^h \approx 0$ , since  $\hat{\phi}_T < \phi$ . It follows that in such cases,  $\hat{e}_{T+h} =$  $(\mu - \hat{\mu}_T) + v_{T+h} + o(\phi^h)$ . Similarly,  $\tilde{e}_{T+h} = -\bar{v}_T + v_{T+h} + o(\phi^h)$ . Hence, for h moderately large and  $\phi$  not too close to the unit circle, a measure of the relative efficiency of the two forecasting approaches can be obtained:

$$\frac{\mathrm{E}(\hat{e}_{T+h}^2)}{\mathrm{E}(\hat{e}_{T+h}^2)} = \frac{\mathrm{E}(\hat{\mu}_T - \mu)^2 + \mathrm{E}(v_{T+h}^2) + o(\phi^h)}{\mathrm{E}(\bar{v}_T^2) + \mathrm{E}(v_{T+h}^2) + o(\phi^h)},\tag{9}$$

since  $(\mu - \hat{\mu}_T)$  and  $\bar{v}_T$  are uncorrelated with  $v_{T+h}$ . But  $E(\mu - \hat{\mu}_T)^2 = O(T^{-1})$ and does not depend on h. To derive  $E(\bar{v}_T^2)$ , recall that  $v_t = \sum_{j=0}^{h-1} \phi^j \varepsilon_{t-j}$ , and hence after some algebra

$$T \ \bar{v}_T = \sum_{j=1}^{h-1} \left(\frac{1-\phi^j}{1-\phi}\right) \phi^{h-j} \varepsilon_{-h+j+1} + \left(\frac{1-\phi^h}{1-\phi}\right) \sum_{t=1}^{T-h+1} \varepsilon_t + \sum_{j=1}^{h-1} \left(\frac{1-\phi^j}{1-\phi}\right) \varepsilon_{T-j+1},$$

and

$$E(\bar{v}_T^2) = Var(\bar{v}_T) = \frac{\sigma_{\varepsilon}^2}{T^2} \left\{ \sum_{j=1}^{h-1} \left( \frac{1-\phi^j}{1-\phi} \right)^2 \left[ 1+\phi^{2(h-j)} \right] + \left( \frac{1-\phi^h}{1-\phi} \right)^2 (T-h+1) \right\}.$$

Clearly,  $E(\bar{v}_T^2) = O(T^{-1})$  if h is fixed. But it is easily seen that we continue to have  $E(\bar{v}_T^2) = O(T^{-1})$  even if  $h \to \infty$  so long as  $h/T \to \kappa$ , where  $\kappa$  is a fixed finite fraction in the range [0, 1). Therefore,

$$\frac{\mathbf{E}(\hat{e}_{T+h}^2)}{\mathbf{E}(\hat{e}_{T+h}^2)} = 1 + O(T^{-1}) + o(\phi^h),$$

and for T sufficiently large there will be little to choose between the iterated and the direct procedures.

This analysis suggests two points. First, we should expect to find the greatest difference between the performance of the two forecasting methods in small samples (T) or in situations where h is large, i.e. when h/T is large. Second, in the context of correctly specified dynamic models there is little reason to choose one approach over the other in large samples or when h/T is small.

To illustrate these results, Figure 1 plots the ratio of RMSE values of the iterated over the direct forecasting approaches for h = 1, 2, ..., 12, assuming  $\phi = 0.9$  and using sample sizes of T = 50, 100, 500, and 1000. The relative efficiency of the iterated approach is very pronounced when T is small or h is large. This is to be expected since the estimation error associated with the direct approach is larger in this situation. Notice also that the advantage of the iterated approach rapidly gets reduced as the sample size is expanded or the forecast horizon is shortened.

In line with the existing literature, the above results conditioned on a given model, namely the AR(1) specification in (1), while ignoring the effect of model misspecification and model uncertainty. It is not clear how to obtain finite-sample analytical results without conditioning on a particular model specification. To deal with this important issue, we next discuss the types of forecasting models under consideration and then turn to a variety of methods for model selection.



## 3 Methods for Multi-period Forecasting

Suppose a forecaster is interested in predicting a  $k \times 1$  vector of target variables  $\mathbf{y}_t = (y_{1t}, y_{2t}, \dots, y_{kt})'$  by means of their own past values and the past values of an additional set of potentially relevant predictor variables,  $\mathbf{x}_t = (x_{1t}, x_{2t}, \dots, x_{Mt})'$ . Typically k is small, often 1 or 2, but M could be large.

For purposes of calculating one-step-ahead forecasts under the iterated approach, the regressors are treated as conditional information and so how they are generated is not a concern. This also holds under the direct forecasting approach irrespective of the forecast horizon. In contrast, when applying the iterated forecasting approach to multi-period horizons, h > 1, the regressors themselves need to be predicted since such values in turn are required to predict future values of y.

#### 3.1 Factor-Augmented VARs

In cases where M is relatively small, say less than ten, one approach is to treat all variables simultaneously, i.e. model  $\mathbf{z}_t = (\mathbf{y}'_t, \mathbf{z}'_t)'$ . Multi-period forecasts of  $y_t$  can then be obtained iteratively using a conventional VAR of the form

$$\boldsymbol{z}_{t} = \boldsymbol{\mu}_{z} + \begin{pmatrix} \boldsymbol{A}_{p}(L) & \boldsymbol{B}_{q}(L) \\ \boldsymbol{C}_{r}(L) & \boldsymbol{D}_{s}(L) \end{pmatrix} \boldsymbol{z}_{t-1} + \boldsymbol{\xi}_{t},$$
(10)

where p and q are the lag order of  $y_t$  and  $x_t$  in the equation for  $y_t$  and r and s is the lag order of  $y_t$  and  $x_t$  in the equation for  $x_t$ .

However, in the common situation where the time-series dimension of the data is limited while M is large, this approach is unlikely to be successful due to the high dimension of the coefficient matrices of the VARs, i.e.  $C_r(L)$  and  $B_q(L)$  in particular.

To deal with this issue, a conditional factor-augmentation approach can be used. In this approach, the large-dimensional  $\boldsymbol{x}_t$ -vector is condensed into a subset of factors,  $\hat{\boldsymbol{f}}_t$ , of dimension m < M, used to summarize the salient features of the larger-dimensional data. A factor-augmented VAR based on the variables  $\tilde{\boldsymbol{z}}_t = (\boldsymbol{y}'_t, \hat{\boldsymbol{f}}'_t)'$  can then be used:

$$\tilde{\boldsymbol{z}}_{t} = \boldsymbol{\mu}_{z} + \begin{pmatrix} \boldsymbol{A}_{p}(L) & \boldsymbol{C}_{q}(L) \\ \boldsymbol{0} & \boldsymbol{D}_{s}^{*}(L) \end{pmatrix} \tilde{\boldsymbol{z}}_{t-1} + \boldsymbol{\xi}_{t}.$$
(11)

Notice the asymmetric treatment of  $y_t$  and  $f_t$  under this approach. Due to the zero in the lower left corner of the lag polynomial matrix pre-multiplying  $\tilde{z}_{t-1}$ , future values of the factors are generated using only current and past values of the factors themselves. The *y*-variables are therefore not used to predict the factors, while the factors are used to predict the *y*-variables.

Only those factors that help predict  $\boldsymbol{y}_t$  are relevant and should be included in this model. This may be a subset of the full set of factors under consideration, which we denote by  $\hat{\boldsymbol{f}}_t^*$ . One then has to choose whether to use the full set of factors  $\hat{\boldsymbol{f}}_t$  to predict the subset of factors,  $\hat{\boldsymbol{f}}_t^*$ , selected when forecasting  $\boldsymbol{y}_{t+h}$ , or whether to use only lagged values of  $\hat{\boldsymbol{f}}_t^*$  to predict their future values. The choice would depend on the number of available factors. In the empirical application below where we consider five factors we shall be using all the factors in forecasting future values of  $\hat{\boldsymbol{f}}_t^*$ .

#### 3.2 Multi-step Forecasts with Dynamic Factor Models

Assume that the  $k \times 1$  vector of target variables,  $\boldsymbol{y}_t$ , is generated according to the following dynamic factor model

$$\boldsymbol{y}_{t} = \boldsymbol{\mu} + \boldsymbol{A}\boldsymbol{y}_{t-1} + \boldsymbol{\Gamma}_{1}\boldsymbol{f}_{t-1} + \boldsymbol{\varepsilon}_{t}, \qquad (12)$$

$$\boldsymbol{\varepsilon}_t \sim \operatorname{iid}(\mathbf{0}, \boldsymbol{\Sigma}),$$

where  $f_t$  is a vector of unobserved common factors, while A,  $\Gamma_1$ , and  $\Sigma$  are unknown coefficient matrices.

In order to forecast  $y_{T+h}$  given information at time T we also need to forecast the factors. Despite the dynamic nature of the above model, we follow Stock and Watson (2002) and estimate  $f_t$  by the principal component (PC) procedure, although one could equally employ the dynamic factor approach of Forni et al. (2005). The key question when using factor models is the choice of the number of factors. This can be determined, for example, using the information criteria (IC) proposed by Bai and Ng (2002). In practice there is a considerable degree of uncertainty surrounding the number of factors to be used, and as pointed out by Bai and Ng (2009), the IC assume that the factors are ordered as predictors of the regressors  $\mathbf{x}_t$ , an ordering that might not be appropriate for predicting the target variables,  $y_t$ . The resultant factors are also often difficult to associate with readily understood economic concepts.

In view of these concerns we adopt a hierarchical approach where we first divide the 170 variables into five economically distinct groups and then select the first PC from each of the categories. All the computations are carried out recursively, so that no future information is used in the construction of the factors. We denote the recursively estimated PC's by  $\hat{f}_t(T-w+1:T)$ , where  $T \geq t$  is the time at which factors are computed and w is the size of the estimation window used to do the computations. To simplify notations we denote  $\hat{f}_t(T-w+1:T)$  by  $\hat{f}_t$ , although strictly speaking both subscripts are needed for appropriate timing when carrying out the computations.<sup>1</sup>

#### **3.3** Iterated System Forecasts

For multivariate forecasts we first select the relevant subset of factors and the lag lengths p and q using the conditional model:

$$y_t = \mu_y + A_p(L)y_{t-1} + C_q(L)f_{t-1} + u_t,$$
 (13)

where

$$A_p(L) = A_0 + A_1 L \dots + A_{p-1} L^{p-1}, C_q(L) = C_0 + C_1 L + \dots + C_{q-1} L^{q-1}.$$

We determine the lag orders p and q and the subset of  $m^*$  factors from the total set of m factors by applying information criteria to the likelihood of  $y_t$ .

For simplicity, we obtain forecasts only from the full set of m factors. Hence we first select a VAR(s) model in  $\hat{f}_t$  so the value of s will be determined by AIC or BIC, again applied recursively. Then

$$\hat{\boldsymbol{f}}_t = \boldsymbol{\mu}_f + \boldsymbol{D}_s(L)\hat{\boldsymbol{f}}_{t-1} + \boldsymbol{\nu}_t, \qquad (14)$$

where

$$\boldsymbol{D}_s(L) = \boldsymbol{D}_0 + \boldsymbol{D}_1 L + \ldots + \boldsymbol{D}_{s-1} L^{s-1}.$$

To compute h-step-ahead forecasts of  $y_t$ , the conditional and marginal models need to be combined:

$$\begin{aligned} y_t &= \mu_y + A_p(L) y_{t-1} + C_q^*(L) f_{t-1} + u_t, \\ \hat{f}_t &= \mu_f + D_s(L) \hat{f}_{t-1} + \nu_t, \end{aligned}$$

<sup>&</sup>lt;sup>1</sup>In the case of an expanding estimation window (with w = T) one would set T = t and work with  $\hat{f}_t(1:t)$ .

or, defining  $\boldsymbol{z}_t = (\boldsymbol{y}_t', \hat{\boldsymbol{f}}_t')'$  and  $\boldsymbol{\xi}_t = (\boldsymbol{u}_t', \boldsymbol{\nu}_t')'$ , we have

$$\boldsymbol{z}_{t} = \begin{pmatrix} \boldsymbol{\mu}_{y} \\ \boldsymbol{\mu}_{f} \end{pmatrix} + \begin{pmatrix} \boldsymbol{A}_{p}(L) & \boldsymbol{C}_{q}^{*}(L) \\ \boldsymbol{0} & \boldsymbol{D}_{s}(L) \end{pmatrix} \boldsymbol{z}_{t-1} + \boldsymbol{\xi}_{t}.$$
(15)

where the selection of factors in the conditional model (13) is reflected in zero-restrictions in  $C_1^*, C_2^*, \ldots C_q^*$ , where the columns corresponding to the factors that are not selected are set to zeros. The factor-augmented VAR in  $z_t$ , (15), can then be iterated forward.<sup>2</sup>

#### 3.4 Univariate Forecasts

Univariate forecasts are a special case of the multivariate forecasts described above. Nevertheless, it is worth briefly clarifying how the forecasts are computed for these models.

For example, with p lags the iterated forecast of  $y_{T+h}$  given information at time T is

$$\hat{y}_{T+h} = \hat{\alpha} + \hat{\beta}_1 \hat{y}_{T+h-1} + \hat{\beta}_2 y_{T+h-2} + \ldots + \hat{\beta}_p y_{T+h-p}$$

where  $\hat{y}_{T+h-1} = y_{T+h-j}$  if  $h \leq j$  and the parameters are estimated from the regression model

$$y_t = \alpha + \beta_1 y_{t-1} + \beta_2 y_{t-2} + \ldots + \beta_p y_{t-p} + \varepsilon_t,$$
  
 $t = T - w + 1 + p, T - w + 2 + p, \ldots, T.$ 

Under the direct approach, for a given lag length, p, the parameters are estimated from the regression

$$y_t = \alpha + \beta_1 y_{t-h} + \beta_2 y_{t-h-1} + \ldots + \beta_p y_{t-h-p} + \varepsilon_t,$$

and the forecast is

$$\hat{y}_{T+h} = \hat{\alpha} + \hat{\beta}_1 y_T + \hat{\beta}_2 y_{T-1} + \ldots + \hat{\beta}_p y_{T-p}.$$

## 4 Model Selection

Given the large dimension of the set of potentially relevant predictor variables, how a particular model is chosen by the forecaster is of great importance. We consider two basic model selection criteria, namely AIC and BIC. Both criteria are commonly used in forecasting studies and have well known properties: AIC achieves a good approximate model as the sample

<sup>&</sup>lt;sup>2</sup>Alternatively, one could model only the factors that have been selected in the conditional model (13). This may be less efficient than using the full VAR in (14), but a smaller number of parameters needs to be estimated from a finite number of observations. It is therefore not clear *a priori* which approach will perform better.

size expands even if the true model is not contained in the universe of models under consideration. However, AIC is not a consistent criterion in the sense that, asymptotically, it does not select the true model with probability one if it happens to be included in the search. In contrast, BIC is a consistent model selection criterion.

The iterated forecasting models can be selected either on the basis of single equations or using a system of equations. For the direct forecasting models a decision has to be made whether or not to correct for the overlap in the observations that affects the sample covariance matrix of the forecast errors, although such corrections will be of second order importance in large samples.

#### 4.1 Iterated Forecasts

Models used to generate the iterated forecasts are selected based on the following criteria. First, we consider applying AIC or BIC to the single equation (k = 1) containing the variable of interest, i.e. for AIC:

AIC = ln[
$$\hat{\boldsymbol{u}}_{i}'\hat{\boldsymbol{u}}_{i}/(T-1)$$
] +  $\frac{2d}{T-1}$ , (16)

where  $\hat{u}_i$  is a  $(T-1) \times 1$  vector of estimated residuals with typical element

$$\hat{u}_{it} = y_{1,t} - \hat{\boldsymbol{\beta}}'_{iT} \boldsymbol{x}_{t-1}, \qquad t = 2, 3, \dots, T,$$

while the  $d \times 1$  vector  $\boldsymbol{x}_{t-1}$  contains subset *i* of  $y_{1t-1}$  and  $\boldsymbol{f}_{t-1}$  and their lags and  $\hat{\boldsymbol{\beta}}_T$  is estimated using observations up to *T*. Similarly, for BIC,

BIC = 
$$\ln[\hat{u}'_i\hat{u}_i/(T-1)] + \frac{\ln(T-1)d}{T-1},$$
 (17)

where  $\hat{\boldsymbol{u}}_i$  is defined as above.

For k > 1, we have

$$AIC = \ln |\hat{\boldsymbol{\Sigma}}_i| + \frac{2kd}{T-1}$$
(18)

where  $\hat{\Sigma}$  is the  $k \times k$  estimated error covariance matrix

$$\hat{\boldsymbol{\Sigma}}_i = \frac{1}{T-1} (\boldsymbol{Y}' \boldsymbol{Y} - \boldsymbol{Y}' \boldsymbol{X}_i (\boldsymbol{X}'_i \boldsymbol{X}_i)^{-1} \boldsymbol{X}'_i \boldsymbol{Y}),$$

 $\boldsymbol{Y}$  is the  $T-1 \times k$  matrix of stacked  $\boldsymbol{y}_t = (y_{1t}, y_{2t}, \dots, y_{kt})'$ , and  $\boldsymbol{X}_i$  is the observation matrix formed from stacking  $\boldsymbol{x}_{it-1}$  over the observations,  $t = 2, 3, \dots, T$ . Similarly,

$$BIC = \ln |\hat{\boldsymbol{\Sigma}}_i| + \frac{\ln(T-1)kd}{T-1}.$$
(19)

#### 4.2 Direct Forecasts

The direct forecasting models are selected based on either AIC

$$AIC = \ln[\hat{\boldsymbol{u}}_h'\hat{\boldsymbol{u}}_h/(T-h)] + \frac{2d}{T-h}$$
(20)

where  $\hat{\boldsymbol{u}}_h$  is a  $(T-h) \times 1$  vector of estimated residuals with typical element

$$\hat{u}_{it} = y_{1t} - \hat{\boldsymbol{\beta}}'_{iT} \boldsymbol{x}_{t-1}, \quad t = h+1, \dots, T,$$

or, using BIC,

BIC = ln[
$$\hat{\boldsymbol{u}}_{h}'\hat{\boldsymbol{u}}_{h}/(T-h)$$
] +  $\frac{\ln(T-h)d}{T-h}$ , (21)

where  $\hat{\boldsymbol{u}}_h$  is defined as above.

For h > 1 the overlap of the forecasts will produce considerable autocorrelation in the residuals. This should be accounted for in small samples when calculating the information criteria. Here we consider two ways to calculate the corrections with further details given in Appendix B. The first approach is to make use of the modified information criteria defined by

$$AIC_{\Pi} = \ln[\hat{\boldsymbol{u}}_{h}'\hat{\boldsymbol{u}}_{h}/(T-h)] + \frac{2\mathrm{tr}(\hat{\boldsymbol{\Pi}}_{S})}{T-h}$$
(22)  
$$BIC_{\Pi} = \ln[\hat{\boldsymbol{u}}_{h}'\hat{\boldsymbol{u}}_{h}/(T-h)] + \frac{\ln(T-h)\mathrm{tr}(\hat{\boldsymbol{\Pi}}_{S})}{T-h},$$

using

$$\mathbf{\hat{\Pi}}_{S} = (\mathbf{X}'\mathbf{X})^{-1}\mathbf{X}'\mathbf{S}\mathbf{X}/h$$

where S is a matrix with h on the diagonal, h-1 on the first diagonal above and below the main diagonal, h-2 on the second diagonal above and below the main diagonal, and so on, i.e.

$$\boldsymbol{S} = \begin{pmatrix} h & h-1 & h-2 & \dots & 0 \\ h-1 & h & h-1 & \dots & 0 \\ h-2 & h-1 & h & & \vdots \\ \vdots & & \ddots & h-1 & h-2 \\ 0 & \dots & & h-1 & h & h-1 \\ 0 & \dots & & h-2 & h-1 & h \end{pmatrix}$$

This formulation aims at capturing the MA(h-1) form of the error process of the overlapping regressions but assumes that the serial correlation of the underlying (non-overlapping) observations is negligible. The second approach uses an estimated covariance matrix. In particular, we use the Newey-West (1987) covariance matrix to obtain the correction. This yields the modified AIC and BIC:

$$AIC_{\tilde{\Pi}} = \ln[\hat{\boldsymbol{u}}_{h}'\hat{\boldsymbol{u}}_{h}/(T-h)] + \frac{2\mathrm{tr}\left(\tilde{\boldsymbol{\Pi}}\right)}{T-h}$$
(23)  
$$BIC_{\tilde{\Pi}} = \ln[\hat{\boldsymbol{u}}_{h}'\hat{\boldsymbol{u}}_{h}/(T-h)] + \frac{\ln(T-h)\mathrm{tr}\left(\tilde{\boldsymbol{\Pi}}\right)}{T-h},$$

where  $\tilde{\mathbf{\Pi}} = \hat{\boldsymbol{\Sigma}}_{xx}^{-1} \hat{\mathbf{\Omega}}, \ \hat{\boldsymbol{\Sigma}}_{xx} = \hat{\sigma}^2 \left(\frac{\mathbf{X}'\mathbf{X}}{T-h}\right), \ \hat{\mathbf{\Omega}}$  is the long-run variance of the residuals as estimated by the Newey-West covariance matrix with bandwidth set to  $\min(h, T^{1/3})$ .

### 5 Monte Carlo Simulations

Having described the forecasting setup and the model selection methods, we next turn to Monte Carlo simulation as a means to evaluate the performance of the various model selection and estimation approaches under two data generating processes (DGPs). Both data generating processes assume the following VAR model for  $y_t = (y_{1t}, y_{2t}, \ldots, y_{kt})'$ :

$$\boldsymbol{y}_t = \boldsymbol{\mu} + \alpha \boldsymbol{A} \boldsymbol{y}_{t-1} + \boldsymbol{H}_{\varepsilon} \boldsymbol{\varepsilon}_t, \quad \boldsymbol{\varepsilon}_t \sim N(\boldsymbol{0}, \boldsymbol{I}_k).$$
(24)

The goodness of fit is controlled by the scaling parameter  $\alpha$ , which is set such that the population  $R^2 = 0.2$  or  $R^2 = 0.8$ , representing low and high predictability scenarios, respectively. An alternative would be to vary  $\mathbf{H}_{\varepsilon}$ . However, for the purpose of controlling the goodness of fit of the equations and given the choices to be considered for  $\mathbf{A}$ , it has proved more convenient to vary  $\alpha$  and set  $\mathbf{H}_{\varepsilon} = \mathbf{I}_k$ . Data are generated for samples of T = 60,120, and 240 observations which in turn are used to generate forecasts for period T + h, so T corresponds to w, the length of a rolling window used for estimation. We consider forecast horizons of h = 1, 3, 6, 12, and 24 periods.

The first DGP assumes that all variables contain useful information for predicting the variable of interest (always the first variable) and so the onestep-ahead forecast should select all variables. Under the second DGP, iterated multi-step forecasts can be expected to be inefficient because they select models that produce good one-step-ahead forecasts and a variable in the DGP is only helpful for longer horizon forecasts.

#### Data Generating Process 1

Under the first DGP, we set k = 3 with  $\mu = 0$ , and choose the *A*-matrix as follows

$$\boldsymbol{A} = \left( \begin{array}{ccc} 0.8 & 0.5 & 0.5 \\ 0 & 0.95 & 0 \\ 0 & 0 & 0.7 \end{array} \right),$$

so that both  $y_2$  and  $y_3$  help predict  $y_1$ , but  $y_{2t}$  and  $y_3$  are in turn not themselves predictable by means of past values of  $y_1$ . Moreover,  $y_2$  is highly persistent while  $y_3$  is not, suggesting that for large values of h in particular,  $y_2$  should play more of a role in forecasting  $y_1$  as compared to  $y_3$ .

#### Data Generating Process 2

The parameters for the second DGP are set to  $k = 4, \mu = 0$ , and

$$oldsymbol{A} = \left( egin{array}{cccccc} 0.8 & 0.5 & 0 & 0 \ 0 & 0.8 & 0.5 & 0 \ 0 & 0 & 0.9 & 0.5 \ 0 & 0 & 0 & 0.95 \end{array} 
ight),$$

Notice that  $y_2$  helps predict  $y_1$ , but  $y_2$  is in turn not itself predictable by means of past values of  $y_1$ . Moreover,  $y_3$  neither predicts nor is predicted by  $y_1$  but  $y_3$  predicts  $y_2$  and therefore may help predict  $y_1$  over medium horizons. Finally, the most persistent variable,  $y_4$ , indirectly helps predict  $y_1$  through its ability to predict  $y_3$ .

#### 5.1 Forecasts

We generate forecasts from both univariate and multivariate models. The univariate forecasts are based on AR models with lag length up to  $p_{\text{max}} = 12$ . The multivariate models consider all regressors in the DGPs with the maximum lag length restricted to  $p_{\text{max}} = 2$ .

In each case forecasts are based on the model selected by one of the criteria discussed in the previous section. Iterated forecasts are then calculated as follows:

$$\hat{\boldsymbol{Z}}_{T+h}^* = \hat{\boldsymbol{\mu}}(\boldsymbol{I} + \hat{\boldsymbol{A}} + \dots + \hat{\boldsymbol{A}}^{h-1}) + \hat{\boldsymbol{A}}^h \boldsymbol{z}_T^*,$$
(25)

where  $\boldsymbol{z}_t^*$  is the sub-vector of  $\boldsymbol{z}_t$  chosen in the model selection step,  $\boldsymbol{Z}_t^* = \boldsymbol{z}_t^*$  if  $\boldsymbol{z}_t^*$  includes  $y_{1t}$  or  $\boldsymbol{Z}_t^* = (y_{1t}, \boldsymbol{z}_t^*)$  if  $\boldsymbol{z}_t^*$  does not include  $y_{1t}$ , and  $\hat{\boldsymbol{\mu}} = \boldsymbol{0}$  if the intercept is not selected.

Direct forecasts are obtained from

$$\tilde{y}_{1,T+h} = \hat{\boldsymbol{\beta}}'_{h} \boldsymbol{x}_{s,T}, \qquad (26)$$

where  $\boldsymbol{x}_{s,T}$  is the subset of regressors that are selected, i.e.,  $\boldsymbol{x}_{s,T} = \boldsymbol{z}_T^*$  if the intercept is not selected or  $\boldsymbol{x}_{s,T} = (1, \boldsymbol{z}_T^*)$  if the intercept is selected.

The forecast errors are calculated as

$$\hat{e}_{T+h} = y_{1,T+h} - \hat{y}_{1,T+h}$$
 or  
 $\tilde{e}_{T+h} = y_{1,T+h} - \tilde{y}_{1,T+h}$  (27)

depending on whether the iterated or direct forecasts are used. Finally, the forecasting performance is measured by the mean squared forecast error (MSFE) computed as

$$MSFE = \frac{1}{R} \sum_{r=1}^{R} e_{T+h,r}^{2}, \qquad (28)$$

where r = 1, 2, ..., R denotes the Monte Carlo replications and  $e_{T+h}$  is either  $\hat{e}_{T+h}$  or  $\tilde{e}_{T+h}$ .

#### 5.2 Summary of Results

Results from the Monte Carlo simulations are reported in Tables 1 and 2. In order to study how the degree of predictability affects the findings, each table contains a panel with  $R^2 = 0.2$  and a panel with  $R^2 = 0.8$ . The former is closer to the empirical results that we obtain, while the second scenario is more relevant for some of the highly persistent variables.

First consider the results for the data generated under DGP 1. For both the univariate models and the VARs, the iterated approach dominates the direct approach. This is a robust finding that holds across sample sizes (T = 60, 120, and 240), information criteria (AIC versus BIC), and forecast horizons (h = 3, 6, 12, and 24). As expected the relative performance of the direct to the iterated forecasts improves with the sample size, T, since it becomes less costly to use an inefficient estimation method in the larger samples.

Turning to a comparison of the univariate and multivariate forecasts, the picture is less clear-cut. In many cases the best overall forecast is generated by the VAR models, although this depends on the degree of predictability assumed for the DGP. Assuming weak predictability ( $R^2 = 0.2$ ), under the iterated approach VAR-models selected by the simple AIC or BIC generally produce better forecasts than their univariate counterparts. This holds across all forecast horizons when T = 60 or T = 120. The performance of the VAR forecasts gets even better relative to the univariate forecasts when the level of predictability is raised by setting  $R^2 = 0.8$ . In this case the VAR-forecasts generate lower MSFE-values than the corresponding univariate forecasts for the vast majority of horizons and sample sizes. These results suggest that the multivariate VAR forecasts perform relatively better when the forecast signal is strong and it becomes easier to correctly identify the relevant predictor variables.

The modifications to the AIC and BIC work well in many cases for  $h \ge 6$  as the data overlap becomes more pronounced. Interestingly, the

band diagonal approach is generally more successful at reducing the MSFEvalues than the Newey-West approach.

Results for the second DGP are presented in Table 2. While the results do not change fundamentally in the case with low predictability  $(R^2 = 0.2)$ , under high predictability  $(R^2 = 0.8)$  the direct forecasting approach applied to the multivariate models now begins to dominate in the larger estimation samples (T = 120 or 240) at the medium-term horizons (h = 3 and 6periods). The iterated approach continues to dominate, however, at the longest horizons (h = 12 and 24) and for the smallest estimation sample (T =60). It also dominates at all horizons for the univariate models. These results are easy to explain since the deterioration in the (relative) performance of the direct forecasting approach can be expected to be greatest when fewer data points are available due either to a small sample size or due to a long horizon which leads to greater estimation errors under this approach.

### 6 Empirical Results

In their analysis, Marcellino, Stock and Watson (2006) (MSW, henceforth) found that iterated univariate forecasts typically outperform direct univariate forecasts. Furthermore, they found that the performance of the iterated univariate forecasts generally improves with the forecast horizon.

MSW studied univariate and bivariate VARs with lag orders either fixed or selected by AIC or BIC. Apart from the search over lag orders, they did not, however, conduct a broad model specification search involving multivariate models. This leaves open the issue of how robust their findings are to a broader model specification search. In this section we consider this question using the same data as in the MSW study.<sup>3</sup> This data set comprises 170 U.S. macroeconomic time series measured at the monthly frequency over the period 1959-2002 (528 months).

#### 6.1 Data Transformations

Following MSW, all variables are transformed by differencing a suitable number of times to achieve stationarity for estimation and model selection. In a second step, forecasts are transformed back to levels and compared to level variables. We briefly explain how the forecasts are computed under the direct and iterated approaches using the autoregressive models as an illustration.

Denote the variables in levels by  $x_t$  and differenced variables by  $y_t$ . AR forecasts can be computed as follows. Under the iterated approach,  $y_{T+h}$  is modelled and  $\hat{x}_{T+h}$  is constructed from  $x_T$  and  $\hat{y}_{T+h}$ . The forecasts of the AR model are based on the sample  $y_{T-w+1}, y_{T-w+2}, \ldots, y_T$ . Here  $y_t = x_t$ 

<sup>&</sup>lt;sup>3</sup>We are grateful to Mark Watson for making this data set publicly available.

if the variable is I(0),  $y_t = \Delta x_t$  if the variable is I(1), and  $y_t = \Delta^2 x_t$  if the variable is I(2), where for each variable we use the same order of integration as MSW.

Under the iterated approach the forecast of  $x_{T+h}$  is constructed from the forecast of  $y_{T+h}$ ,  $x_T$ , and  $x_{T-1}$  as follows

$$\hat{x}_{T+h} = \begin{cases} \hat{y}_{T+h} & \text{if } x_t \text{ is I}(0) \\ x_T + \sum_{i=1}^h \hat{y}_{T+i} & \text{if } x_t \text{ is I}(1) \\ x_T + h\Delta x_T + \sum_{i=1}^h \sum_{j=1}^i \hat{y}_{T+j} & \text{if } x_t \text{ is I}(2) \end{cases}$$

Similarly, under the direct approach, the forecast of  $x_{T+h}$  is constructed from the forecast of  $y_{T+h}$ ,  $x_T$ , and  $x_{t-1}$  as follows

$$\hat{x}_{T+h} = \begin{cases} \hat{y}_{T+h} & \text{if } x_t \text{ is I}(0) \\ x_T + \hat{y}_{T+h} & \text{if } x_t \text{ is I}(1) \\ x_T + h\Delta x_T + \hat{y}_{T+h} & \text{if } x_t \text{ is I}(2) \end{cases}$$

#### 6.2 Setup

Forecasting is performed recursively and begins in 1979M1 (with a minimum of w observations before forecasting), running until the end of the sample 2002M12. This yields up to 288 forecasts for h = 1, 286 for h = 3, and so on. No allowance is made for the possibility of structural breaks. Forecasts are again reported for horizons of h = 1, 3, 6, 12, and 24 months. Two window lengths are used for estimation, namely w = 120 and w = 240, that is, 10 and 20 years of data. Fixing the window length allows us to better understand the relative performance of the various model selection criteria which can be quite sensitive to the sample size.

To address the effect of model selection on the multivariate forecasts, we extract factors from the 170 series in five groups, namely (A) one factor for "income, output, sales, capacity utilization" (38 variables); (B) one factor for "employment and unemployment" (27 variables); (C) one factor for "construction, inventories, and orders" (37 variables); (D) one factor for "interest rates and asset prices" (33 variables); and (E) one factor for "nominal prices, wages, and money" (35 variables).

To avoid any look-ahead biases, as noted earlier the factors are estimated recursively, i.e.  $f_{it} = f(\mathbf{X}_i(1:t))$ . We then obtain forecasts of the factors from VARs fitted to all five factors with lag length chosen by AIC or BIC and  $p_{\text{max}} = 2$ .

The search over multivariate models is thus conducted over specifications that include own lags as well as factors. The space of models is limited as follows. For the univariate autoregressive models the possible lag lengths are p = 0, 1, 2, ..., 12, where p = 0 is an intercept only model. For the factor-augmented VAR models we search across five factors with one or two lags in addition to an intercept. For computational simplicity the lag length is set to be the same for  $y_{it}$  and  $f_{it}$ .

#### 6.3 Forecasting Performance

Empirical results are presented in Tables 3-8. We first show MSFE results averaged across all 170 variables (Table 3), then present evidence on how many variables get included under the different approaches (Table 4) and report the proportion of cases for which the univariate iterated approach based on models selected by AIC produces lower MSFE-values than the alternative forecasting methods (Table 5). We also present test results from formal model comparisons (Table 6). To gain further insights into the forecasting performance for different types of economic variables, Tables 7 and 8 present MSFE values and test results for subcategories of the variables described above.

#### 6.3.1 Univariate Models

Table 3 shows the relative forecasting performance (measured by MSFE) of the iterated and direct methods averaged over all 170 variables included in the MSW data set. For the shortest estimation window (w = 120), the iterated forecasts are better on average than the direct ones, particularly at long horizons (h = 12 and 24 months) and when using AIC. To understand this note that the iterated forecasts require a smaller pre-sample for the estimation and therefore tend to have a lower parameter estimation error. Such errors are most important in large models (AIC penalizes large models less than the BIC) and when the sample size is short. The relatively more efficient use of data by the iterated approach also explains why this performs better than the direct approach even with the large sample size (w = 240) when selecting models by the AIC.

In the large estimation sample (w = 240), the finding that the iterated forecasts produce the best performance is overturned across all forecast horizons when the BIC is used as the model selection criterion. This is partly explained by the fact that BIC chooses fewer parameters and so estimation error is less of an issue.

The modified information criteria, in particular the band-diagonal approach, are generally quite successful in improving the average forecasting performance relatively to using the simple version of the AIC. This finding holds less frequently when the BIC is used to select models.

In the short sample (w = 120), the best overall short-term (h = 1 or 3) forecasting performance is produced by the models selected by the BIC. At all other horizons, the univariate iterated models selected by the AIC deliver the best average performance. In the larger sample (w = 240), however, the best forecasting performance for horizons of 3 and 6 months is produced by the direct forecast approaches. The AIC modified to using a band-diagonal covariance matrix does particularly well. Once again the univariate iterated approach based on the AIC dominates overall when h = 12 and 24.

Table 4 shows the number of predictor variables included by the different approaches, averaged across time and across target variables. Univariate models selected by the AIC generally include 3-4 variables in the small sample and 4-5 variables in the larger sample. For the univariate models selected by the BIC, this number declines to only 1-2 variables in the small sample and two variables on average in the larger sample.

The average MSFE-values reported in Table 3 may be dominated by the most volatile variables and could provide an incomplete picture of relative forecasting performance. To deal with this, Table 5 shows the proportion of the 170 variables for which the iterated univariate AR forecasts based on models selected by the AIC generate a larger MSFE than the various alternatives. We use the iterated univariate forecasts selected by the AIC as our benchmark given the earlier evidence that this approach is the best univariate model, a finding supported by the results reported by MSW.

Among the univariate forecasts, in the small sample (w = 120), only the short-term iterated forecasts based on models selected by the BIC produce a majority of cases that outperform the iterated AIC. In the longer sample (w = 240), however, the iterated AIC approach is clearly best on average, particularly at the long horizons.

#### 6.3.2 Multivariate Models

Turning to the multivariate models, the factor-augmented iterated VAR forecasts that use models selected by the AIC produce better forecasts than their univariate counterparts for a majority of the variables. This holds across all horizons and in both sample sizes. However, the factor-augmented iterated VAR forecasts that use models selected by the BIC produce the best forecasts for most variables at the four shortest horizons when w = 120 and for two horizons, h = 1 and h = 12, when w = 240.

Table 3 also shows that in many cases the proposed modifications to the information criteria continue to lead to improved performance for the direct forecasting models.

Table 4 shows that, as expected, the multivariate models include more predictor variables than their univariate counterparts. Under the AIC, on average 5-7 regressors get included in the small sample, rising to 6-8 variables in the larger sample. Once again, the BIC leads to somewhat smaller models with 3-4 predictor variables.

#### 6.4 Model Comparisons

Table 6 provides test results based on a formal comparison of the benchmark iterated AIC model with the alternative approaches listed in each row. To this end we adopt the test methodology advocated by Giacomini and White (2006) which is ideally suited for our purpose since we are conducting pairwise model comparisons and use rolling window estimators. The table lists the proportion of model comparisons (out of 170) for which the null of equal predictive accuracy is rejected against the alternative that the two models' accuracy differ at a particular significance level.

Excluding the zeros which arise when forecasting methods are essentially identical, the proportion of rejections ranges between 0.15 and 0.40, with values near 0.15 and 0.20 most common. Asterisks indicate when the iterated AR model with lags selected by the AIC produces a smaller MSFE in fewer than 50% or 25% of the cases where the null of equal predictive accuracy is rejected.

The test results provide statistical evidence that on average the univariate iterated AIC approach performs very well relative to the other univariate methods and there is little evidence on which to prefer alternative univariate methods.

Among the factor-augmented VAR models, the iterated forecasts do better on average than the direct forecasts when a short estimation window (w = 120) is used. These results are overturned, however, when the long estimation window (w = 240) is used. In the latter case, the direct forecasting method is better for most forecast horizons irrespective of which information criterion is used.

For the models selected by the AIC or BIC, in many cases the Newey-West and band-diagonal modifications to the covariance matrix help improve on the average performance of the direct forecasts.

Overall, the factor models tend to generate smaller average forecast errors than the best univariate approach for the short horizons, h = 1 and h = 3, but do worse than the iterated AIC when h = 6, 12 and 24 months. Our results thus suggest that including more variables favors the direct method at short forecast horizons.

#### 6.5 Results by Variable Categories

The empirical results turn out to be quite similar for four of the five categories of economic variables, namely (A) income, output, sales and capacity utilization, (B) employment and unemployment, (C) Construction, inventories and orders, and (D) interest rates and asset prices. In contrast, quite different results are obtained for the fifth category, namely (E) Nominal prices, wages and money. For this reason, Table 7 presents results averaged across variables in categories A-D, while Table 8 shows separate results for category-E variables.

Table 7 shows that the benefit from using the multivariate factor-based approach comes out very strongly for the first four categories. For these variables, across the vast majority of sample sizes and forecast horizons, the multivariate iterated AIC produces lower MSFE-values than the univariate iterated AIC approach. Moreover, the modifications to the AIC or BIC methods applied to the multivariate models often generate the best overall forecasting performance.

Across the first four categories of variables, the multivariate iterated approach based on the AIC generates the best results on average. In fact, for these variables, the multivariate iterated AIC approach dominates the univariate iterated AIC across all forecast horizons and for both long and short estimation samples. The multivariate iterated approach based on BIC also performs well, particularly for the shortest estimation window (w = 120).

Table 8 shows that it is only for the final group of variables, (E) nominal prices, wages and money, that the univariate iterated AIC approach performs best on average.

Overall, these findings clearly demonstrate the value from utilizing multivariate information and also provide evidence that our proposed refinements to the information criteria work in many cases.

## 7 Conclusion

Our empirical results show an interesting interaction between the length of the estimation window, how strongly a particular model selection method penalizes the inclusion of additional variables, the forecast horizon, and the relative performance of the direct versus iterated approaches. Estimation error turns out to be key to understanding our findings. The iterated forecasting approach performs well relative to the direct approach when the sample size is small, when using an information criterion such as the AIC that does not penalize additional parameters too heavily and when the forecast horizon gets large. This is what we would expect to find since the iterated approach makes more efficient use of sample data than the direct forecasting approach. In this regard the theoretical results obtained for the simple AR(1) model also provide useful insights. Conversely, when the BIC is used, leading to smaller models and reduced parameter estimation error, the direct approach outperforms the iterated approach on average.

We confirm the finding reported by Marcellino, Stock and Watson (2006) that iterated forecasts generated by univariate models selected by the AIC produce good forecasts on average at long horizons. However, we also find that multivariate VAR-based forecasts outperform the univariate forecasts at short horizons (h = 1 and 3 months) and across all horizons for four of the five categories of economic variables studied here. This suggests that, for most economic variables and particularly at short horizons, information beyond what is contained in the past history of the variables themselves can be helpful in producing better forecasts.

## Appendix: Mathematical details

### Appendix A: Proof of Proposition 1

**Direct forecast** First consider the direct method, and note that from equation (3)

$$\bar{y}_T = a_h + \phi_h \bar{y}_{T,-h}, + \bar{v}_T$$

which together with equation (7) yields

$$(a_h - \tilde{a}_{h,T}) + (\phi_h - \tilde{\phi}_{h,T})\bar{y}_{T,-h} + \bar{v}_T = 0.$$
 (A1)

Note that the forecast error for the direct method is given by

$$\tilde{e}_{T+h} = (a_h - \tilde{a}_{h,T}) + (\phi_h - \bar{\phi}_{h,T})y_T + v_{T+h},$$

which, using (A1), can be written as

$$\tilde{e}_{T+h} = (\phi_h - \tilde{\phi}_{h,T})(y_T - \bar{y}_{T,-h}) - \bar{v}_T + v_{T+h}.$$
(A2)

But using equation (2), we have

$$(y_T - \bar{y}_{T,-h}) = u_T - \bar{u}_{T,-h},$$

where in a similar way  $\bar{u}_{T,-h} = T^{-1} \sum_{t=1}^{T} u_{t-h}$ . Also from equation (8)

$$\tilde{\phi}_{h,T} - \phi_h = \frac{\sum_{t=1}^T (y_{t-h} - \bar{y}_{T,-h}) v_t}{\sum_{t=1}^T (y_{t-h} - \bar{y}_{T,-h})^2} \\ = \frac{\sum_{t=1}^T (u_{t-h} - \bar{u}_{T,-h}) v_t}{\sum_{t=1}^T (u_{t-h} - \bar{u}_{T,-h})^2},$$

and

$$\tilde{e}_{T+h} = (u_T - \bar{u}_{T,-h}) \left( \frac{\sum_{t=1}^T (u_{t-h} - \bar{u}_{T,-h}) v_t}{\sum_{t=1}^T (u_{t-h} - \bar{u}_{T,-h})^2} \right) - \bar{v}_T + v_{T+h}.$$

Recall that  $v_t$  and  $u_t$  are linear functions of the  $\varepsilon_t$  shocks. It follows that  $\tilde{e}_{T+h}$  is an odd function of  $u_t$  and  $v_t$ , and so long as the errors are symmetrically distributed around zero, we have  $E(\tilde{e}_{T+h}) = 0$ , assuming that T is sufficiently large so that expectations exists.

**Iterated forecasts** Consider next the forecast errors of the iterated procedure and note that

$$\hat{e}_{T+h} = a\left(\frac{1-\phi^h}{1-\phi}\right) - \hat{a}_T\left(\frac{1-\hat{\phi}_T^h}{1-\hat{\phi}_T}\right) + \left(\phi^h - \hat{\phi}_T^h\right)y_T + v_{T+h},$$

which we re-write as

$$\hat{e}_{T+h} = -(\hat{\mu}_T - \mu)(1 - \hat{\phi}_T^h) - (\hat{\phi}_T^h - \phi^h) u_T + v_{T+h},$$

where  $\mu = a/(1-\phi)$  as before and  $\hat{\mu}_T = \hat{a}_T/(1-\hat{\phi}_T)$ . But

$$\hat{a}_{T} - a = \hat{\mu}_{T} - \mu - (\hat{\mu}_{T}\hat{\phi}_{T} - \mu\phi) = \hat{\mu}_{T} - \mu - (\hat{\mu}_{T} - \mu)\hat{\phi}_{T} - \mu(\hat{\phi}_{T} - \phi),$$
(29)

and from equation (6) we also have

$$a - \hat{a}_T + (\phi - \hat{\phi}_T) \ \bar{y}_{h:T,-1} + \bar{\varepsilon}_{h:T} = 0,$$

where  $\bar{\varepsilon}_{h:T} = (T+h-1)^{-1} \sum_{t=-h+2}^{T} \varepsilon_t$ . Using this result in (29) we have

$$(\hat{\phi}_T - \phi) \ \bar{y}_{h:T,-1} - \bar{\varepsilon}_{h:T} = (\hat{\mu}_T - \mu) \left(1 - \hat{\phi}_T\right) - \mu(\hat{\phi}_T - \phi),$$

from which it follows that

$$\hat{\mu}_{T} - \mu = \frac{(\hat{\phi}_{T} - \phi) (\bar{y}_{h:T,-1} - \mu) - \bar{\varepsilon}_{h:T}}{1 - \hat{\phi}_{T}}$$

$$= \frac{(\hat{\phi}_{T} - \phi) \bar{u}_{h:T,-1} - \bar{\varepsilon}_{h:T}}{1 - \hat{\phi}_{T}}$$

$$= \frac{(\hat{\phi}_{T} - \phi) \bar{u}_{h:T,-1} - \bar{\varepsilon}_{h:T}}{1 - \phi - (\hat{\phi}_{T} - \phi)},$$

where  $\bar{u}_{h:T,-1} = (T+h-1)^{-1} \sum_{t=-h+2}^{T} u_{t-1}$ . The small sample bias of  $\hat{\phi}_T$  ensures that  $1 - \hat{\phi}_T \neq 0$ . Hence

$$\hat{e}_{T+h} = (\mu - \hat{\mu}_T)(1 - \phi^h) + \left(\hat{\phi}_T^h - \phi^h\right)(\hat{\mu}_T - \mu) - \left(\hat{\phi}_T^h - \phi^h\right)u_T + v_{T+h}.$$

Now let  $\hat{\delta}_T = \hat{\phi}_T - \phi$  and note that

$$\hat{\phi}_T^h - \phi^h = \sum_{j=1}^h \binom{h}{j} \hat{\delta}_T^j \phi^{h-j},$$

and

$$\hat{\delta}_T = \frac{\sum_{t=-h+2}^T (u_{t-h} - \bar{u}_{T,-1}) v_t}{\sum_{t=-h+2}^T (u_{t-h} - \bar{u}_{T,-1})^2}.$$

Therefore, it follows that since  $\hat{\delta}_T$  is an even function of  $u_t$  and  $v_t$ , then  $\hat{\phi}_T^h - \phi^h$  will also be an even function of the underlying shocks. Furthermore, since

$$\hat{\mu}_T - \mu = \frac{\delta_T \ \bar{u}_{h:T,-1} - \bar{\varepsilon}_{h:T}}{1 - \phi - \hat{\delta}_T},$$

then  $\hat{\mu}_T - \mu$  is also an odd function of the underlying shocks. Hence  $\hat{e}_{T+h}$  will be an odd function of the shocks and assuming its expectation exists we have  $E(\hat{e}_{T+h}) = 0$  as required.

Specifically

$$\hat{e}_{T+h} = -\left(\frac{\hat{\delta}_T \ \bar{u}_{h:T,-1} - \bar{\varepsilon}_{h:T}}{1 - \phi - \hat{\delta}_T}\right) (1 - \phi^h) + \left(\sum_{j=1}^h \binom{h}{j} \hat{\delta}_T^j \phi^{h-j}\right) \left(\frac{\hat{\delta}_T \ \bar{u}_{h:T,-1} - \bar{\varepsilon}_{h:T}}{1 - \phi - \hat{\delta}_T} - u_T\right) + v_{T+h}.$$

#### Appendix B: Derivation of modified information criteria

Consider the *h*-step ahead forecast in (3) and for simplicity of exposition assume that  $\alpha_h = 0$  and drop the subscript from the parameter, so that we have

$$\boldsymbol{y}_{t+h} = \boldsymbol{\phi} \boldsymbol{y}_t + \boldsymbol{v}_t. \tag{B1}$$

Denote the loss evaluated at  $\phi$  by  $\ell(\phi)$ , while  $\ell_0(\phi) = \mathbb{E}[\ell(\phi)]$ , and the *j*th derivative of the loss is given by  $\ell^{(j)}(\phi)$ . Moreover, let  $\phi_0$  be the true parameter and  $\hat{\phi}$  its estimate. Using these notations, we have that  $\ell_0^{(1)}(\phi_0) = 0$  and  $\ell^{(1)}(\hat{\phi}) = 0$ . For  $\ell(\phi) = \frac{1}{2\sigma^2} (\boldsymbol{Y}_{T+h} - \boldsymbol{Y}_T \phi)' (\boldsymbol{Y}_{T+h} - \boldsymbol{Y}_T \phi)$  we also have that  $\ell^{(2)}(\phi) = \ell_0^{(2)}(\phi_0) = \frac{1}{2\sigma^2} \boldsymbol{Y}'_T \boldsymbol{Y}_T$ , where  $\boldsymbol{Y}_{T+h} = (\boldsymbol{y}_{1+h}, \boldsymbol{y}_{2+h}, \dots, \boldsymbol{y}_{T+h})'$  and  $\boldsymbol{Y}_T = (\boldsymbol{y}_1, \boldsymbol{y}_2, \dots, \boldsymbol{y}_T)'$ , which implies that a second order Taylor expansion is exact.

Define  $\ell_0(\hat{\phi})$  as the loss incurred by using the estimated parameter  $\hat{\phi}$  instead of the unknown true parameter. Then

$$\mathrm{E}[\ell_0(\hat{\boldsymbol{\phi}})] = \mathrm{E}[\ell(\hat{\boldsymbol{\phi}})] + \mathrm{E}[\ell_0(\hat{\boldsymbol{\phi}}) - \ell(\hat{\boldsymbol{\phi}})].$$

A second order Taylor expansion implies

$$\ell_0(\hat{\phi}) = \ell_0(\phi_0) + (\hat{\phi} - \phi_0)' \ell_0^{(2)}(\phi_0)(\hat{\phi} - \phi_0)$$

and

$$\ell(\hat{\phi}) = \ell(\phi_0) + (\hat{\phi} - \phi_0)' \ell^{(1)}(\phi_0) + (\hat{\phi} - \phi_0)' \ell^{(2)}_0(\phi_0)(\hat{\phi} - \phi_0) \; .$$

Taking expectations of these expressions yields

$$\begin{split} \mathrm{E}[\ell(\hat{\phi})] &= \mathrm{E}[\ell(\phi_0)] + \mathrm{E}[(\hat{\phi} - \phi_0)'\ell^{(1)}(\phi_0)] + \mathrm{E}[(\hat{\phi} - \phi_0)'\ell_0^{(2)}(\phi_0)(\hat{\phi} - \phi_0)] \\ &= \ell_0(\phi_0) + \mathrm{E}[(\hat{\phi} - \phi_0)'\ell^{(1)}(\phi_0)] + \mathrm{E}[(\hat{\phi} - \phi_0)'\ell_0^{(2)}(\phi_0)(\hat{\phi} - \phi_0)], \end{split}$$

and therefore

$$E[\ell_0(\hat{\phi}) - \ell(\hat{\phi})] = -E[(\hat{\phi} - \phi_0)'\ell^{(1)}(\phi_0)].$$

Furthermore, by Taylor expansion we have

$$(\hat{\phi} - \phi_0) = -[\ell^{(2)}(\phi_0)]^{-1}\ell^{(1)}(\phi_0),$$

and therefore

$$\begin{aligned} -\mathrm{E}[(\hat{\boldsymbol{\phi}} - \boldsymbol{\phi}_0)'\ell^{(1)}(\boldsymbol{\phi}_0)] &= -\mathrm{E}[\mathrm{tr}(\sigma^{-2}\boldsymbol{U}_T'\boldsymbol{Y}_T(\boldsymbol{Y}_T'\boldsymbol{Y}_T)^{-1}\boldsymbol{Y}_T\boldsymbol{U}_T)] \\ &= -\mathrm{tr}(\sigma^{-2}(\boldsymbol{Y}_T'\boldsymbol{Y}_T)^{-1}\boldsymbol{Y}_T'\mathrm{E}[\boldsymbol{U}_T'\boldsymbol{U}_T]\boldsymbol{Y}_T), \end{aligned}$$

where  $U_T = (v_1, v_2, ..., v_T)$ .

If the errors  $v_t$  were iid, this would give the standard penalty term K. However, in overlapping forecasts the errors will be autocorrelated and the expression will not collapse to K.

## References

- [1] Bai, Jushan and Serena Ng, 2002, Determining the number of factors in approximate factor models. *Econometrica*, 70, 161–221.
- [2] Bai, Jushan and Serena Ng, 2009, Boosting diffusion indices, *Journal* of Applied Econometrics, forthcoming.
- [3] Bernanke, Ben S., Jean Boivin and Piotr Eliasz, 2005, Measuring the effect of monetary policy: A factor-augmented vector autoregressive (FAVAR) approach. *Quarterly Journal of Economics* 120, 387–422.
- [4] Bhansali, Rajendra J., 1999, Parameter estimation and model selection for multistep prediction of a time series: a review. in Asymptotics, Non-parametrics and Time Series (ed. Subir Ghosh), 201–225, Marcel Dekker, New York.
- [5] Brown, Bryan W. and Roberto S. Mariano, 1989, Measures of deterministic prediction bias in nonlinear models. *International Economic Review* 30(3), 667–684.
- [6] Clements, Michael, and David Hendry, 1998, Forecasting Economic Time Series. Cambridge University Press, Cambridge.
- [7] Cox, David R., 1961, Prediction by exponentially weighted moving averages and related methods. *Journal of the Royal Statistical Society* B 23, 414–422.
- [8] Findley, David F., 1983, On the use of multiple models for multi-period forecasting. Proceedings of Business and Economic Statistics, American Statistical Association, 528–531.

- [9] Forni Mario, Hallin Marc, Lippi Marco, Reichlin Lucrezia, 2005, The generalized dynamic factor model, one sided estimation and forecasting. *Journal of the American Statistical Association*, 100, 830–840.
- [10] Fuller, Wayne A., 1996, Introduction to Statistical Time Series. Wiley, New York.
- [11] Giacomini, Raffaella and Halbert White, 2006, Tests of conditional predictive ability. *Econometrica* 74(6), 1545–1578.
- [12] Ing, Ching-Kang, 2003, Multistep prediction in autoregressive processes. *Econometric Theory* 19, 254–279.
- [13] Marcellino, Massimiliano, James H. Stock and Mark W. Watson, 2006, A comparison of direct and iterated multistep AR methods for forecasting macroeconomic time series. *Journal of Econometrics* 135, 499–526.
- [14] Newey, Whitney K. and Mark W. West, 1987, A simple, positive semidefinite, heteroskedasticity and autocorrelation consistent covariance matrix. *Econometrica* 55(3), 703–708.
- [15] Pesaran, M. Hashem, and Allan Timmermann, 2005, Small sample properties of forecasts from autoregressive models under structural breaks. *Journal of Econometrics* 129, 183–217.
- [16] Phillips, Peter C. B., 1979, The sampling distribution of forecasts from a first order autoregression. *Journal of Econometrics*, 9, 241–261.
- [17] Schorfheide, Frank, 2005, VAR forecasting under misspecification. Journal of Econometrics, 128, 99–136.
- [18] Stock, James H., and Mark W. Watson, 2002, Forecasting Using Principal Components from a Large Number of Predictors. *Journal of the American Statistical Association*, 97, 1167–1179.
- [19] Stock, James H., and Mark W. Watson (2005) 'Implications of dynamic factor models for VAR Analysis.' *mimeo*.
- [20] Ullah, Aman, 2004, *Finite Sample Econometrics*. Oxford University Press, Oxford.

| $\begin{array}{ c c c c c c c c c c c c c c c c c c c$   | 87         1.0618           93         1.0618           25         1.0618           21         1.0618           22         1.0618           21         1.0564           14         1.0564           10551         1.0564           25         1.0564           1.0564         1.0564           25         1.0564           1.0564         1.0302           92         1.0302           92         1.0302           92         1.0302           92         1.0302           93         1.0302  | $\begin{array}{c ccccccccccccccccccccccccccccccccccc$   | $\begin{array}{c} R, \ T=60\\ 1.1499 \ 1\\ 1.1881 \ 1\\ 1.1881 \ 1\\ 1.1881 \ 1\\ 1.1741 \ 1\\ 1.1741 \ 1\\ 1.1518 \ 1\\ 1.1419 \ 1\\ 1.1518 \ 1\\ 1.1618 \ 1\\ 1.1794 \ 1\\ 1.1794 \ 1\\ 1.1619 \ 1\\ 1.1659 \ 1\\ 1.1659 \ 1\\ 1.1669 \ 1\\ 1.1661 \ 1\\ 1.1669 \ 1\\ 1.1661 \ 1\\ 1.1668 \ 1\\ 1\\ 1.1668 \ 1\\ 1\\ 1.1668 \ 1\\ 1.1668$  | $\begin{array}{c c} A \\ \hline 1.1595 \\ 1.1595 \\ 1.1678 \\ 1.1416 \\ 1.1445 \\ 1.1445 \\ 1.1445 \\ 1.1445 \\ 1.1456 \\ 1.1450 \\ 1.1855 \\ 1.1855 \\ 1.1813 \\ 1.1813 \\ 1.1813 \\ 1.1813 \\ 1.1813 \\ 1.1813 \\ 1.1701$                                     |  |
|--|---|---|--|---|--|
| $ \begin{array}{ c c c c c c c c c c c c c c c c c c c$  | 87         1.0618           33         1.0618           25         1.0618           21         1.0618           21         1.0618           21         1.0618           22         1.0654           14         1.0564           25         1.0564           25         1.0564           26         1.0564           27         1.0564           28         1.0564           29         1.0302           92         1.0302           92         1.0302           92         1.0302           93         1.0302           93         1.0302           93         1.0302 | $ \begin{bmatrix} 501 & 1.13\\ 2042 & 1.27\\ 2075 & 1.27\\ 2075 & 1.41\\ 1439 & 1.13\\ 1562 & 1.17\\ 1562 & 1.17\\ 1562 & 1.21\\ 1562 & 1.21\\ 1564 & 1.21\\ 1584 & 1.21\\ 1881 & 1.19\\ 1881 & 1.19\\ 1884 & 1.20\\ 1$   |  | $\begin{array}{c} 1.1499 & 1.1\\ 1.1814 & 1.5\\ 1.1881 & 1.5\\ 1.1741 & 1.5\\ 1.1376 & 1.1\\ 1.1376 & 1.1\\ 1.1376 & 1.1\\ 1.1419 & 1.1\\ 1.1518 & 1.1\\ 1.1247 & 1.1\\ 1.1749 & 1.5\\ 1.1749 & 1.5\\ 1.1749 & 1.5\\ 1.1749 & 1.5\\ 1.1749 & 1.5\\ 1.1618 & 1.1\\ 1.1618 & $   | $\begin{array}{cccccccccccccccccccccccccccccccccccc$               |
| $ \begin{array}{ c c c c c c c c c c c c c c c c c c c$  | $\begin{array}{c ccccccccccccccccccccccccccccccccccc$   | $\begin{array}{c ccccccccccccccccccccccccccccccccccc$   | <u>i ci ci     </u>  | $\begin{array}{c} 1.1241 & 1.2\\ 1.1741 & 1.2\\ 1.1376 & 1.1\\ 1.1376 & 1.1\\ 1.1518 & 1.1\\ 1.1518 & 1.1\\ 1.1518 & 1.1\\ 1.1247 & 1.1\\ 1.1749 & 1.2\\ 1.1749 & 1.2\\ 1.1749 & 1.2\\ 1.1749 & 1.2\\ 1.1749 & 1.2\\ 1.1618 & 1.1\\ 1.1618 & $   | $\begin{array}{cccccccccccccccccccccccccccccccccccc$               |
| $ \begin{array}{ c c c c c c c c c c c c c c c c c c c$  | 21         1.0618           42         1.0564           114         1.0564           02         1.0551           25         1.0564           32         1.0564           92         1.0302           92         1.0302           93         1.0302           93         1.0302           93         1.0302  | $\begin{array}{c ccccccccccccccccccccccccccccccccccc$   |  | $\begin{array}{c} 1.1741 & 1.2\\ 1.1376 & 1.1\\ 1.1447 & 1.1\\ 1.1518 & 1.10\\ 1.1518 & 1.10\\ 1.1518 & 1.10\\ 1.1518 & 1.10\\ 1.1247 & 1.11\\ 1.1794 & 1.20\\ 1.1817 & 1.20\\ 1.1817 & 1.20\\ 1.1817 & 1.20\\ 1.1671 & 1.18\\ 1.1659 & 1.18\\ 1.1659 & 1.18\\ 1.1659 & 1.18\\ 1.1659 & 1.18\\ 1.1659 & 1.18\\ 1.1659 & 1.18\\ 1.1659 & 1.18\\ 1.1659 & 1.18\\ 1.1659 & 1.18\\ 1.1659 & 1.18\\ 1.1659 & 1.18\\ 1.1659 & 1.18\\ 1.1650 & 1.18\\ 1.$  | $\begin{array}{cccccccccccccccccccccccccccccccccccc$               |
| $\begin{array}{c c c c c c c c c c c c c c c c c c c $   | $\begin{array}{c ccccccccccccccccccccccccccccccccccc$   | $\begin{array}{cccccccccccccccccccccccccccccccccccc$  | $\begin{array}{c c} 1.126\\ \hline 1.156\\ \hline 1.156\\ \hline 1.162\\ \hline 1.162\\ \hline 1.184\\ \hline 1.1$ | $\begin{array}{c} 1.1447 & 1.156\\ 1.1447 & 1.156\\ 1.1518 & 1.164\\ 1.1518 & 1.163\\ 1.1247 & 1.137\\ 1.1794 & 1.204\\ 1.1749 & 1.204\\ 1.1749 & 1.208\\ 1.1659 & 1.188\\ 1.1659 & 1.188\\ 1.1671 & 1.186\\ 1.1671 & 1.186\\ 1.1867 & 1.186\\ 1.1867 & 1.186\\ 1.1867 & 2.3697 & 2.458\\ 2.3697 & 2.458\\ 2.3697 & 2.458\\ 2.3697 & 2.458\\ 1.186\\$ | $\begin{array}{cccccccccccccccccccccccccccccccccccc$               |
| $ \begin{array}{c ccccccccccccccccccccccccccccccccccc$   | 02         1.0551           25         1.0564           32         1.0302           42         1.0302           97         1.0299           92         1.0364           1.0378  | $\begin{array}{c ccccccccccccccccccccccccccccccccccc$   |  | $\begin{array}{c} 1.1518 & 1.1646\\ 1.1419 & 1.1622\\ \hline 1.1419 & 1.1622\\ 1.1247 & 1.1376\\ 1.1794 & 1.2046\\ 1.1817 & 1.2086\\ 1.1749 & 1.2016\\ 1.1817 & 1.2086\\ 1.1671 & 1.1346\\ 1.1659 & 1.1889\\ 1.1671 & 1.1898\\ 1.1618 & 1.1888\\ 1.1618 & 1.1888\\ 1.1844 & 1.1888\\ 1.1618 & 1.1888\\ 1.1618 & 1.1888\\ 1.1618 & 1.1888\\ 1.1618 & 1.1888\\ 1.1618 & 1.1888\\ 1.1618 & 1.1888\\ 1.1618 & 1.1888\\ 1.1618 & 1.1888\\ 1.1618 & 1.1888\\ 1.1618 & 1.1888\\ 1.1618 & 1.1888\\ 1.1844 & 1.1844 & 1.1888\\ 1.1844 & 1.1844 & 1.1844\\ 1.1844 & 1.1844 & 1.1844 & 1.1844\\ 1.1844 & 1.1844 & 1.1844 & 1.1844 & 1.1844 & 1.1844 & 1.1844 & 1.1844 & 1.1844 &$  | $\begin{array}{cccccccccccccccccccccccccccccccccccc$               |
| $ \begin{array}{ c c c c c c c c c c c c c c c c c c c$  | 25         1.0564           32         1.0302           42         1.0302           97         1.0299           92         1.0364           93         1.0365           93         1.0378   | $\begin{array}{c c} 1.29\\ 1.129\\ 1.212\\ 1.216\\ 1.216\\ 1.216\\ 1.206\\ 1.196\\ 1.196\\ 1.196\\ 1.196\\ 1.196\\ 1.206\\ 1$   |  | $\begin{array}{c ccccccccccccccccccccccccccccccccccc$   | $\begin{array}{c ccccccccccccccccccccccccccccccccccc$              |
| $ \begin{array}{ c c c c c c c c c c c c c c c c c c c$  | 32         1.0302           42         1.0302           97         1.0299           92         1.0302           99         1.0364           90         1.0378   | 1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.22/<br>1.19/<br>1.19/<br>1.19/<br>1.19/<br>1.19/<br>1.19/<br>1.20/<br>1.20/<br>1.20/<br>1.20/<br>1.20/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.21/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/<br>1.22/ | .1379<br>.2040<br>.2040<br>.2011<br>.1346<br>.1881<br>.1881<br>.1898<br>.1844<br>.1844<br>.1844  | $\begin{array}{c ccccccccccccccccccccccccccccccccccc$   | $\begin{array}{c ccccccccccccccccccccccccccccccccccc$              |
| $ \begin{array}{ c c c c c c c c c c c c c c c c c c c$  | 42 1.0302<br>97 1.0299<br>92 1.0302<br>99 1.0364<br>70 1.0378   | 1.21<br>1.21<br>1.21<br>1.21<br>1.22<br>1.19<br>1.19<br>1.19<br>1.19<br>1.19<br>1.19<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.2000<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200<br>1.200  | 2040<br>2011<br>2011<br>2011<br>2011<br>2011<br>2011<br>2011   | $\begin{array}{cccccccccccccccccccccccccccccccccccc$  | $\begin{array}{cccccccccccccccccccccccccccccccccccc$               |
| $ \begin{array}{ c c c c c c c c c c c c c c c c c c c$  | $\begin{array}{c c} 97 & 1.0299 \\ 92 & 1.0302 \\ 99 & 1.0364 \\ 70 & 1.0378 \end{array}$   | 1.219<br>1.229<br>1.119<br>1.119<br>1.119<br>1.119<br>1.119<br>1.119<br>1.1200<br>2.410   | .2086<br>.2011<br>.1346<br>.1881<br>.1881<br>.1898<br>.1898<br>.1844<br>.1844  | $\begin{array}{cccccccccccccccccccccccccccccccccccc$  | $\begin{array}{cccccccccccccccccccccccccccccccccccc$               |
| $ \begin{array}{ c c c c c c c c c c c c c c c c c c c$  | $\begin{array}{c c} 92 & 1.0302 \\ 99 & 1.0364 \\ 70 & 1.0378 \\ \end{array}$   | $\begin{array}{c c}1.22\\1.119\\1.19\\1.19\\1.19\\2.410\\2.410\\2.410\\2.60\\2.60\\2.60\\2.60\\2.60\\2.60\\2.60\\2.6$   | .2011<br>.1346<br>.1881<br>.1881<br>.1898<br>.1898<br>.1844<br>.1844   | $\begin{array}{cccccccccccccccccccccccccccccccccccc$  | $\begin{array}{cccccccccccccccccccccccccccccccccccc$               |
| $\begin{array}{c ccccccccccccccccccccccccccccccccccc$  | $99 \mid 1.0364$<br>70 \ 1.0378   | $\begin{array}{c} 1.119\\1.19\\1.19\\1.19\\1.19\\2.41\\0\\2.41\\0\\2.80\\0\\2.80\\0\\2.80\\0\\2.80\\0\\0\\2.80\\0\\0\\0\\0\\0\\0\\0\\0\\0\\0\\0\\0\\0\\0\\0\\0\\0\\0\\$   | .1346<br>.1881<br>.1898<br>.1898<br>.1844<br>.1844   | $\begin{array}{cccccccccccccccccccccccccccccccccccc$  | $\begin{array}{cccccccccccccccccccccccccccccccccccc$               |
| $\begin{array}{c c c c c c c c c c c c c c c c c c c $   | 70 + 1 + 0.378  | $\begin{array}{c c} 1.19' \\ 1.192 \\ 1.200 \\ 2.410 \\ 2.410 \\ 2.80'$  | 1881<br>1898<br>1844<br>1844<br>4584   | $\begin{array}{cccccccccccccccccccccccccccccccccccc$  | $\begin{array}{cccccccccccccccccccccccccccccccccccc$               |
| $\begin{array}{ c c c c c c c c c c c c c c c c c c c$   |   | $\begin{array}{c c} 1.193 \\ 1.200 \\ 2.410 \\ 2.410 \\ 2.300$  | .1898<br>.1844<br>.1584<br>.4584   | $\begin{array}{cccccccccccccccccccccccccccccccccccc$  | $\begin{array}{cccccccccccccccccccccccccccccccccccc$               |
| $\begin{array}{ c c c c c c c c c c c c c c c c c c c$   | $\begin{array}{c ccccccccccccccccccccccccccccccccccc$   | 0.41  | .4584 2  | $\begin{array}{c} R, T = 60 \\ 2.3697 & 2.4584 \\ 2.3697 & 2.253 \\ 2.2697 & 2.257 \\ 2.2$  | $\begin{array}{c} AR,T=60\\ 2.0168  2.3697  2.4584  2 \end{array}$ |
| AR, $T = 120$ AR, $T = 240$ 2463       1.9314       2.2907       2.3418       2.3145       1.2455       1.8920       2.2142       2.3643        2463       1.9577       2.3480       2.4557       1.2455       1.9004       2.2399       2.3243        2463       1.9577       2.3480       2.4557       1.2455       1.9009       2.2425       2.3176        2463       1.9599       2.3613       2.4435       2.4529       1.2455       1.9009       2.2425       2.3176        2340       1.9148       2.7119       2.3354       2.3132       1.2415       1.8890       2.2092       2.3877        2361       1.9148       2.3724       2.3132       1.2415       1.8890       2.2092       2.2897        2361       1.9369       2.3256       2.3447       1.2415       1.8890       2.2974        2340       1.9369       2.3256       2.3444       1.2415       1.8890       2.2455       2.2854        2341       1.9369       2.3256       2.3444       1.2415       1.8890       2.2414       2.2897        2340       1.9369       2.3256       2.3444       1.2415       1.8891 | _   | .41(  | . 4584 2<br>8518 3   | $\begin{array}{c} R,  T = 60 \\ \hline 2.3697 & 2.4584 & 2 \\ \hline 2.2717 & 2.0710 & 2 \\ \hline \end{array}$   | AR, T = 60<br>2.0168 2.3697 2.4584 2                               |
| $\begin{array}{ c c c c c c c c c c c c c c c c c c c$   |   | 41(   | .4584 2.<br>8518 3   | 2.3697 2.4584 2.  | 2.0168 $2.3697$ $2.4584$ $2.$                                      |
| $\begin{array}{ c c c c c c c c c c c c c c c c c c c$   | 66   1.2463   | òc  | 8518 3   | 0 111 0 0110 0  |  |
| $\begin{array}{ c c c c c c c c c c c c c c c c c c c$   | 70   1.2463   | 0.00  |  |   | 2.0812 2.5517 2.8518   |
| $\begin{array}{ c c c c c c c c c c c c c c c c c c c$   | 56   1.2470   | 4<br>2<br>2<br>2  | .8046 3  | 2.5044 2.8646 3   | 2.1055 2.5044 2.8046 3   |
| $\begin{array}{ c c c c c c c c c c c c c c c c c c c$   | $34   1.2403 \\ 44   1.2340$  | 0.04<br>0.32  | . / 801  | 2.3213 2.4153   | 2.0942 $2.0320$ $2.101$  |
| $\begin{array}{ c c c c c c c c c c c c c c c c c c c$   | 82 1.2340   | 3.00  | .6450  | 2.4285 $2.6450$   | 2.0295 $2.4285$ $2.6450$   |
| $\begin{array}{ c c c c c c c c c c c c c c c c c c c$   | 36   1.2361   | 3.18  | .6760  | 2.4457 $2.6760$   | 2.0422 $2.4457$ $2.6760$   |
| $VAR, T = 120 \qquad VAR, T = 240$ 0240 1.6578 2.1116 2.2718 2.3016 1.0136 1.6107 2.0772 2.28550240 1.6856 2.1782 2.3928 2.4481 1.0136 1.6210 2.1031 2.33410255 1.6917 2.1816 2.3867 2.4392 1.0135 1.6217 2.1038 2.33290240 1.6998 2.1799 2.3708 2.4176 1.0136 1.6238 2.1059 2.32980195 1.6552 2.1105 2.2712 2.3020 1.0107 1.6081 2.0761 2.2854  | 33  1.2340  | 3.22;   | .5287  | 2.3886 $2.5287$   | 2.0893 2.3886 2.5287   |
| $\begin{array}{c ccccccccccccccccccccccccccccccccccc$  |   |   |  | 4R, T = 60  | VAR, T = 60  |
| $\begin{array}{c ccccccccccccccccccccccccccccccccccc$  | 52   1.0240   | 2.35!   | .3834  | 2.2013 $2.3834$   | 1.7210 $2.2013$ $2.3834$   |
| $\begin{array}{c ccccccccccccccccccccccccccccccccccc$  | $53 \mid 1.0240$  | 2.84  | .6873  | 2.3706 $2.6873$   | 1.7919 $2.3706$ $2.6873$   |
| $\begin{array}{c ccccccccccccccccccccccccccccccccccc$  | $18 \mid 1.0255$  | 2.85  | .6866  | 2.3691 $2.6866$   | 1.8018 2.3691 2.6866   |
| .0195 $1.6552$ $2.1105$ $2.2712$ $2.3020$ $1.0107$ $1.6081$ $2.0761$ $2.2854$  | 63   1.0240   | 2.83(   | .6432  | 2.3692 $2.6432$   | 1.8214 $2.3692$ $2.6432$   |
|  | 22   1.0195   | 2.35;   | .3783  | 2.1948 2.3783   | 1.7173 2.1948 2.3783   |
| .0195 $1.6978$ $2.1711$ $2.3696$ $2.4232$ $1.0107$ $1.6194$ $2.1022$ $2.3259$  | 05   1.0195   | 2.79(   | .6447  | 2.3436  2.6447  | 1.8104 $2.3436$ $2.6447$   |
| .0193 $1.7073$ $2.1773$ $2.3604$ $2.4144$ $1.0108$ $1.6226$ $2.1064$ $2.3235$  | 60   1.0193   | 2.79(   | .6344  | 2.3509 	2.6344  | 1.8166  2.3509  2.6344   |
| $.0195  1.7466  2.1914  2.3473  2.3858  \left  \begin{array}{ccc} 1.0107 & 1.6338 & 2.1200 & 2.3193 \end{array} \right  \\ \end{array}$  | 39   1.0195   | 2.74:   | .5743  | 2.3610 $2.5743$   | 1.8727 $2.3610$ $2.5743$   |

| Table 3: Relative for 170 minutes for the former of the fo | orecasting<br>Marcoll | g perform    | hance of i    | terated a   | nd direct   | methods<br>sot | s: Averag    | se ratio of    | : MSFE a    | cross all |
|--|-----------------------|--------------|---------------|-------------|-------------|----------------|--------------|----------------|-------------|-----------|
|  |                       |              |               | nz) IInent  | uu) uara    | 200            |              |                |             |           |
| Forecast horizon   | 1                     | 3            | 6             | 12          | 24          | 1              | 3            | 6              | 12          | 24        |
|  |                       | A            | R, w = 1      | 20          |             |                | A            | $R, w = 2^{i}$ | 40          |           |
| iterated AIC   | 1.0000                | 1.0000       | 1.0000        | 1.0000      | 1.0000      | 0.9809         | 0.9629       | 0.9470         | 0.9546      | 1.0018    |
| direct AIC   | 1.0000                | 1.0152       | 1.0178        | 1.0821      | 1.2234      | 0.9809         | 0.9605       | 0.9455         | 0.9887      | 1.1083    |
| mod. $AIC(NW)$   | 0.9932                | 1.0025       | 1.0138        | 1.0844      | 1.2450      | 0.9738         | 0.9537       | 0.9411         | 0.9858      | 1.1115    |
| mod. AIC(diag)   | 1.0000                | 1.0124       | 1.0170        | 1.1049      | 1.2113      | 0.9809         | 0.9602       | 0.9451         | 0.9833      | 1.1219    |
| iterated BIC   | 0.9892                | 1.0100       | 1.0368        | 1.0529      | 1.0460      | 0.9876         | 1.0090       | 1.0459         | 1.0906      | 1.1413    |
| direct BIC   | 0.9892                | 1.0080       | 1.0162        | 1.0766      | 1.1731      | 0.9876         | 0.9674       | 0.9539         | 0.9900      | 1.1012    |
| mod. BIC(NW)   | 0.9908                | 1.0065       | 1.0217        | 1.0832      | 1.1821      | 0.9892         | 0.9755       | 0.9573         | 0.9995      | 1.1050    |
| mod. BIC(diag)   | 0.9892                | 1.0083       | 1.0296        | 1.1739      | 1.2218      | 0.9876         | 0.9659       | 0.9544         | 1.0265      | 1.1942    |
|  |                       | Fac          | tors, $w =$   | 120         |             |                | Fac          | tors, $w =$    | 240         |           |
| iterated AIC   | 0.9855                | 0.9822       | 1.0397        | 1.0821      | 1.0658      | 0.9684         | 0.9743       | 1.0302         | 1.0597      | 1.0951    |
| direct AIC   | 0.9855                | 0.9866       | 1.0500        | 1.1310      | 1.2829      | 0.9684         | 0.9464       | 0.9981         | 1.0510      | 1.1428    |
| mod. AIC(NW)   | 0.9838                | 0.9827       | 1.0554        | 1.1411      | 1.2871      | 0.9675         | 0.9465       | 0.9968         | 1.0510      | 1.1446    |
| mod. AIC(diag)   | 0.9855                | 0.9867       | 1.0462        | 1.1146      | 1.2517      | 0.9684         | 0.9443       | 0.9938         | 1.0358      | 1.1131    |
| iterated BIC   | 0.9775                | 0.9810       | 1.0272        | 1.0642      | 1.1149      | 0.9694         | 0.9883       | 1.0502         | 1.0986      | 1.1556    |
| direct BIC   | 0.9775                | 0.9815       | 1.0221        | 1.1076      | 1.2606      | 0.9694         | 0.9564       | 1.0002         | 1.0452      | 1.1264    |
| mod. BIC(NW)   | 0.9729                | 0.9825       | 1.0356        | 1.1187      | 1.2729      | 0.9698         | 0.9671       | 1.0085         | 1.0383      | 1.1289    |
| mod. BIC(diag)   | 0.9775                | 0.9866       | 1.0294        | 1.1595      | 1.3241      | 0.9694         | 0.9562       | 0.9947         | 1.0523      | 1.1958    |
| The table reports th   | e MSFE c              | of the diffe | rent foreca   | sts as a ra | atio of the | MSFE of        | the iterat   | ed AR fore     | ecast based | on the    |
| model selected by Al   | [C with $w$           | = 120, wh    | ere $w$ is th | e length of | f the estim | ation wind     | low. AR re   | esults are b   | ased on u   | iivariate |
| autoregressive model   | s and resu            | lts labeled  | 'Factors' a   | re based o  | n multivar  | iate factor    | -augmente    | d VAR mo       | dels. The l | MSFE is   |
| calculated only for the  | nose foreca           | st periods   | where fore    | casts from  | all metho   | ds are ava     | ilable. The  | IV. Pom, a     | C(NW)' an   | pom, p    |
| BIC(NW)' forecasts a   | are based o           | in the mod   | ified AIC a   | nd BIC, wl  | here the co | variance m     | atrix is est | imated usi     | ng the New  | ey-West   |
| estimator; the 'mod.   | AIC(diag)             | om, pue '(   | d. BIC(dia    | g)' forecas | ts are base | ed on the i    | nodified A   | IC and BI      | C, where t  | he band   |
| diagonal covariance 1  | natrix is u           | sed.         |               |             |             |                |              |                |             |           |

| Lade 4:  | Average                    | number                             | or param                   | eters incl              | iuaea by                    | the vario                 | us ioreca                  | sung app                  | roacnes                   |                    |
|--|----------------------------|------------------------------------|----------------------------|-------------------------|-----------------------------|---------------------------|----------------------------|---------------------------|---------------------------|--------------------|
| forecast horizon   | 1                          | ۍ                                  | 9                          | 12                      | 24                          |                           | e.                         | 6                         | 12                        | 24                 |
|  |                            | A                                  | R, w = 1                   | 20                      |                             |                           | $A_{-}$                    | R, w = 24                 | 40                        |                    |
| iterated AIC   | 3.6818                     | 3.6739                             | 3.6643                     | 3.6528                  | 3.6268                      | 5.3329                    | 5.3324                     | 5.3268                    | 5.3097                    | 5.2644             |
| direct AIC   | 3.6818                     | 3.8997                             | 3.6890                     | 3.5925                  | 3.6599                      | 5.3329                    | 5.4674                     | 4.9338                    | 4.7061                    | 4.5515             |
| mod. AIC(NW)   | 3.5544                     | 3.8351                             | 3.8012                     | 3.8965                  | 4.3633                      | 4.5663                    | 4.7830                     | 4.5292                    | 4.4382                    | 4.5739             |
| mod. AIC(diag)   | 3.6818                     | 3.9146                             | 3.6884                     | 3.1524                  | 2.6761                      | 5.3329                    | 5.4836                     | 4.9307                    | 4.4363                    | 3.3783             |
| iterated <b>BIC</b>  | 1.5772                     | 1.5761                             | 1.5750                     | 1.5749                  | 1.5722                      | 2.0762                    | 2.0741                     | 2.0708                    | 2.0671                    | 2.0564             |
| direct BIC   | 1.5772                     | 1.6242                             | 1.6727                     | 1.5076                  | 1.0992                      | 2.0762                    | 2.5768                     | 2.5133                    | 2.1432                    | 1.4824             |
| mod. BIC(NW)   | 1.4430                     | 1.5965                             | 1.7039                     | 1.6698                  | 1.4564                      | 1.7677                    | 2.1656                     | 2.3122                    | 2.0125                    | 1.5315             |
| mod. BIC(diag)   | 1.5772                     | 1.6596                             | 1.6283                     | 0.9403                  | 0.6774                      | 2.0762                    | 2.6165                     | 2.5194                    | 1.7473                    | 0.9544             |
|  |                            | Fac                                | tors, $w =$                | 120                     |                             |                           | Fact                       | fors, $w =$               | 240                       |                    |
| iterated AIC   | 5.6457                     | 5.6479                             | 5.6524                     | 5.6632                  | 5.6738                      | 6.7936                    | 6.7982                     | 6.8044                    | 6.8130                    | 6.8327             |
| direct AIC   | 5.6457                     | 6.2418                             | 6.6317                     | 7.1535                  | 7.4837                      | 6.7936                    | 7.4745                     | 7.7807                    | 8.2464                    | 8.6145             |
| mod. AIC(NW)   | 5.5718                     | 5.9080                             | 6.3140                     | 6.9700                  | 7.3650                      | 6.4328                    | 6.6605                     | 6.9104                    | 7.3851                    | 7.9594             |
| mod. AIC(diag)   | 5.6457                     | 5.5882                             | 5.3345                     | 5.3025                  | 4.9973                      | 6.7936                    | 6.7860                     | 6.3810                    | 6.1154                    | 5.6833             |
| iterated <b>BIC</b>  | 2.9691                     | 2.9697                             | 2.9714                     | 2.9765                  | 2.9852                      | 3.6786                    | 3.6838                     | 3.6902                    | 3.6995                    | 3.7182             |
| direct BIC   | 2.9691                     | 3.5073                             | 3.8466                     | 4.4481                  | 4.7807                      | 3.6786                    | 4.3675                     | 4.6880                    | 5.2442                    | 5.6612             |
| mod. BIC(NW)   | 2.8732                     | 3.2199                             | 3.5118                     | 4.2826                  | 4.7391                      | 3.2386                    | 3.6748                     | 3.7356                    | 4.3888                    | 4.8309             |
| mod. BIC(diag)   | 2.9691                     | 2.9437                             | 2.7650                     | 2.5562                  | 2.0400                      | 3.6786                    | 3.8031                     | 3.5475                    | 3.2871                    | 2.4881             |
| The table reports the and across all $170 v_{\rm a}$<br>Table 3. | e average r<br>ariables in | umber of <sub>I</sub><br>the Marce | barameters<br>llino et al. | chosen by<br>(2006) dat | the differen<br>ta set. For | at model set<br>the model | selection prc<br>selection | cedures, av<br>methods se | veraged ov<br>e the footı | er time<br>note to |

| approach    |  |
|-------------|--|
| forecasting |  |
| the various |  |
| by t        |  |
| included    |  |
| parameters  |  |
| fo .        |  |
| number      |  |
| Average     |  |
| Table 4:    |  |

| Table 5: Forecasti                           | ing perfo                 | rmance n     | neasured 1   | relative t  | o the ite    | rated AR    | forecast   | ing mode                | l selected                | by AIC         |
|--|---------------------------|--------------|--------------|-------------|--------------|-------------|------------|-------------------------|---------------------------|----------------|
| Forecast horizon                             | 1                         | 3            | 9            | 12          | 24           | 1           | 3          | 9                       | 12                        | 24             |
|  |                           | A            | R, w = 1     | 20          |              |             | A          | $R, w = 2^{\iota}$      | 40                        |                |
| direct AIC                                   |                           | 0.4353       | 0.4294       | 0.2294      | 0.1412       |             | 0.5294     | 0.5118                  | 0.3647                    | 0.1765         |
| mod. $AIC(NW)$                               | 0.5059                    | 0.5235       | 0.4529       | 0.2471      | 0.1235       | 0.5118      | 0.5471     | 0.4941                  | 0.3706                    | 0.2059         |
| mod. AIC(diag)                               |                           | 0.4235       | 0.4000       | 0.2412      | 0.1471       |             | 0.5529     | 0.5118                  | 0.3706                    | 0.1824         |
| iterated <b>BIC</b>                          | 0.6294                    | 0.5294       | 0.5294       | 0.4765      | 0.4941       | 0.5118      | 0.5059     | 0.4059                  | 0.3765                    | 0.4059         |
| direct BIC                                   | 0.6294                    | 0.4235       | 0.4353       | 0.2176      | 0.1824       | 0.5118      | 0.4412     | 0.4235                  | 0.3235                    | 0.1882         |
| mod. $BIC(NW)$                               | 0.5706                    | 0.4118       | 0.4294       | 0.2471      | 0.1765       | 0.4824      | 0.4176     | 0.3824                  | 0.2882                    | 0.1941         |
| mod. BIC(diag)                               | 0.6294                    | 0.4588       | 0.4235       | 0.2353      | 0.1765       | 0.5118      | 0.4471     | 0.4353                  | 0.2647                    | 0.1529         |
|  |                           | Fac          | tors, $w =$  | 120         |              |             | Fac        | tors, $w =$             | 240                       |                |
| iterated AIC                                 | 0.6118                    | 0.6176       | 0.5353       | 0.5294      | 0.5059       | 0.6588      | 0.6471     | 0.6000                  | 0.5647                    | 0.5941         |
| direct AIC                                   | 0.6118                    | 0.5647       | 0.4412       | 0.3412      | 0.2059       | 0.6588      | 0.6647     | 0.5294                  | 0.4176                    | 0.4118         |
| mod. AIC(NW)                                 | 0.6235                    | 0.5706       | 0.4118       | 0.2941      | 0.2000       | 0.6824      | 0.6529     | 0.5294                  | 0.4294                    | 0.4294         |
| mod. AIC(diag)                               | 0.6118                    | 0.5706       | 0.4353       | 0.3588      | 0.2588       | 0.6588      | 0.6529     | 0.5588                  | 0.4824                    | 0.4353         |
| iterated <b>BIC</b>                          | 0.6529                    | 0.7059       | 0.6118       | 0.5412      | 0.4588       | 0.6824      | 0.6118     | 0.5647                  | 0.5706                    | 0.5647         |
| direct BIC                                   | 0.6529                    | 0.6235       | 0.4941       | 0.3412      | 0.2000       | 0.6824      | 0.6529     | 0.5353                  | 0.4235                    | 0.4235         |
| mod. $BIC(NW)$                               | 0.6647                    | 0.6118       | 0.4882       | 0.3235      | 0.2235       | 0.6824      | 0.6353     | 0.5176                  | 0.4588                    | 0.4353         |
| mod. BIC(diag)                               | 0.6529                    | 0.5941       | 0.5176       | 0.3059      | 0.2706       | 0.6824      | 0.6353     | 0.5353                  | 0.4824                    | 0.4294         |
| The table reports the<br>a. larger MSFE than | e proportio<br>the foreca | on of variat | oles for whi | ch the iter | ated AR for  | precasts ba | sed on the | model sele<br>above 0.5 | cted by AI<br>0 suggest 1 | C has<br>hat a |
| particular method de                         | ominates th               | ne univaria  | te iterated  | AIC forec   | asts for a n | najority of | the variab | les. For the            | e model sel               | ection         |
| methods see the foot                         | note to Ta                | ble 3.       |              |             |              |             |            |                         |                           |                |

| ٦I                      |   |
|-------------------------|---|
| 2                       |   |
| <u>6</u> 1              |   |
|                         |   |
| ĕ                       |   |
| 히                       |   |
| el P                    |   |
| - Sel                   |   |
| <u> </u>                |   |
| ല                       |   |
| 81                      |   |
| Ξl                      |   |
| -                       |   |
| 쮑                       |   |
| ٠E١                     |   |
| $\overline{\mathbf{S}}$ |   |
| 8                       |   |
| el                      |   |
| . <u></u>               |   |
| -                       |   |
| щI                      | 1 |
|                         | 1 |
| 7                       | 1 |
| Ъ                       | 1 |
| 5                       |   |
| e                       |   |
| :::                     |   |
| υ                       |   |
| <u>-</u> []             | ŀ |
| 7                       |   |
| 님                       |   |
| e                       |   |
| <u>ا ۲</u> .            |   |
| Et l                    |   |
| 픵                       |   |
| ЩЦ                      |   |
| -7                      |   |
| el                      |   |
| 31                      |   |
| Sel                     |   |
| ĕ                       |   |
| 8                       |   |
| ان                      | 1 |
| ۲ <u>ۆ</u>              | 1 |
| ar                      | 1 |
| Π                       |   |
| ΞI                      |   |
| 요                       |   |
| Ч                       |   |
| Ă                       | 1 |
| 60                      |   |
| . <u> </u>              | 1 |
| ١Ę                      | 1 |
| a l                     | 1 |
| SI                      |   |
| ΞI                      |   |
| Ы                       | 1 |
| · .                     | 1 |
| n)                      | 1 |
| <u>e</u>                | 1 |
| إي                      |   |
| $ \mathbf{a} $          | 1 |
| L '                     | ľ |

|                                  |              | Table                    | e 6: Pairw   | rise mode   | l compari   | ison tests   |              |               |                |              |
|----------------------------------|--------------|--------------------------|--------------|-------------|-------------|--------------|--------------|---------------|----------------|--------------|
| Forecast horizon                 | 1            | 3                        | 9            | 12          | 24          | 1            | 3            | 9             | 12             | 24           |
|                                  | AR  fd       | recasts                  | against      | iterated    |             | AR fo        | recasts      | against       | iterated       |              |
|                                  | AR(AI0       | $\mathcal{I}$ , $w = 12$ | 50           |             |             | AR(AIC       | y, w = 24    | 0             |                |              |
| direct AIC                       |              | 0.1588                   | 0.1765       | 0.2118      | 0.3588      |              | $0.1176^{*}$ | $0.1824^{*}$  | 0.1647         | 0.1588       |
| mod. AIC(NW)                     | $0.1647^{*}$ | 0.1588                   | 0.1471       | 0.2118      | 0.3176      | $0.1588^{*}$ | $0.1353^{*}$ | $0.1353^{*}$  | 0.1235         | 0.1647       |
| mod. AIC(diag)                   |              | 0.1647                   | 0.1824       | 0.2765      | 0.3176      |              | $0.1176^{*}$ | $0.1706^{*}$  | 0.1765         | 0.1824       |
| iterated BIC                     | $0.1529^{*}$ | 0.2235                   | 0.2412       | 0.1529      | 0.1706      | 0.2294       | 0.2941       | 0.2824        | 0.2353         | 0.1941       |
| direct BIC                       | $0.1529^{*}$ | 0.1765                   | 0.1765       | 0.2647      | 0.3353      | 0.2294       | 0.2059       | 0.1235        | 0.1235         | 0.1471       |
| mod. BIC(NW)                     | 0.1471       | 0.2412                   | 0.2000       | 0.2353      | 0.3176      | 0.2059       | 0.2647       | 0.1471        | 0.1882         | 0.1588       |
| mod. BIC(diag)                   | $0.1529^{*}$ | 0.1824                   | 0.2294       | 0.4118      | 0.3000      | 0.2294       | 0.2059       | 0.1412        | 0.2235         | 0.2294       |
|                                  | Factor       | fore casts               | against      | iterated    |             | Factor       | fore casts   | against       | iterated       |              |
|                                  | AR(AH        | C), w = 12               | 50           |             |             | AR(AIC       | y, w = 24    | 0             |                |              |
| iterated AIC                     | $0.1706^{*}$ | $0.2294^{*}$             | 0.2294       | 0.1765      | 0.1706      | $0.2294^{*}$ | 0.2353       | 0.2176        | 0.1765         | $0.1882^{*}$ |
| direct AIC                       | $0.1706^{*}$ | 0.1647                   | 0.1412       | 0.1294      | 0.1353      | $0.2294^{*}$ | 0.2000       | 0.1706        | 0.1471         | 0.1706       |
| mod. $AIC(NW)$                   | $0.1824^{*}$ | $0.1412^{*}$             | 0.1529       | 0.1471      | 0.1529      | $0.2353^{*}$ | $0.1941^{*}$ | 0.1471        | 0.1588         | 0.1706       |
| mod. AIC(diag)                   | $0.1706^{*}$ | $0.1471^{*}$             | 0.1471       | 0.1118      | 0.1294      | $0.2294^{*}$ | 0.1882       | 0.1588        | 0.1529         | 0.1353       |
| iterated BIC                     | $0.1824^{*}$ | 0.2412                   | 0.2059       | 0.1941      | 0.1882      | $0.2353^{*}$ | 0.2529       | 0.2824        | 0.2235         | 0.1824       |
| direct BIC                       | $0.1824^{*}$ | 0.1647                   | 0.1706       | 0.1118      | 0.1235      | $0.2353^{*}$ | $0.2353^{*}$ | 0.1765        | 0.1588         | 0.1471       |
| mod. BIC(NW)                     | $0.1647^{*}$ | 0.1706                   | 0.1647       | 0.1235      | 0.1412      | $0.2765^{*}$ | $0.2529^{*}$ | 0.1647        | 0.1294         | 0.1529       |
| mod. BIC(diag)                   | $0.1824^{*}$ | 0.1588                   | 0.1294       | 0.1647      | 0.2059      | $0.2353^{*}$ | $0.2647^{*}$ | 0.1882        | 0.2235         | 0.2176       |
| The table reports the            | e results of | applying the             | e Giacomini  | and White   | (2006)  tes | t to the nul | l that the f | precast prec  | ision of the i | iterated     |
| AR based on AIC ar               | nd the meth  | nod in the re            | espective ro | w are ident | ical. We sl | now the pro  | portion of   | rejections of | the null at    | the $5\%$    |
| level. $*$ (**) indicate         | s that the i | iterated AR              | based on $P$ | AIC has a s | maller MS.  | FE in less t | han 50% (:   | 55%) of the   | forecasts. 7   | The test     |
| statistics are calcula           | ted only for | r those peri             | ods where f  | orecasts fr | om all met  | hods are av  | zilable. Fo  | r the mode    | l selection n  | nethods      |
| see the footnote to <sup>¬</sup> | lable 3.     |                          |              |             |             |              |              |               |                |              |

| Table 7: Relative for             | orecasting 1    | performanc     | e among va    | riables in (  | Category (    | A)–(D) (av    | verages ove   | r 135 variał   | oles)              |              |
|-----------------------------------|-----------------|----------------|---------------|---------------|---------------|---------------|---------------|----------------|--------------------|--------------|
| Forecast horizon                  | 1               | 3              | 9             | 12            | 24            | 1             | 3             | 9              | 12                 | 24           |
|                                   |                 | A              | R, w = 120    | (             |               |               | A             | R, w = 240     | (                  |              |
| iterated AIC                      | 1.0000          | 1.0000         | 1.0000        | 1.0000        | 1.0000        | 0.9776        | 0.9652        | 0.9470         | 0.9457             | 0.9720       |
| direct AIC                        | 1.0000          | 1.0230         | 1.0255        | 1.0928        | 1.2061        | 0.9776        | $0.9647^{*}$  | 0.9499         | 0.9841             | 1.0726       |
| mod. $AIC(NW)$                    | $0.9915^{*}$    | 1.0070         | 1.0200        | 1.0944        | 1.2252        | $0.9680^{*}$  | $0.9555^{*}$  | $0.9435^{*}$   | 0.9805             | 1.0771       |
| mod. AIC(diag)                    | 1.0000          | 1.0197         | 1.0244        | 1.1187        | 1.1781        | 0.9776        | $0.9644^{*}$  | 0.9497         | 0.9786             | 1.0837       |
| iterated <b>BIC</b>               | $0.9809^{**}$   | $0.9844^{*}$   | $0.9915^{*}$  | $0.9944^{*}$  | $0.9868^{*}$  | $0.9695^{*}$  | 0.9611        | 0.9560         | $0.9560^{*}$       | $0.9731^{*}$ |
| direct BIC                        | $0.9809^{**}$   | 0.9975         | 1.0088        | 1.0721        | 1.1385        | $0.9695^{*}$  | 0.9595        | 0.9490         | 0.9731             | 1.0396       |
| mod. BIC(NW)                      | $0.9822^{*}$    | 0.9976         | 1.0150        | 1.0819        | 1.1461        | $0.9719^{*}$  | 0.9613        | 0.9525         | 0.9819             | 1.0456       |
| mod. BIC(diag)                    | $0.9809^{**}$   | 0.9994         | 1.0238        | 1.1721        | 1.1082        | $0.9695^{*}$  | 0.9583        | 0.9511         | 1.0172             | 1.0692       |
|                                   |                 | Fac            | tors, $w = 1$ | 20            |               |               | Fac           | tors, $w = 2$  | 40                 |              |
| iterated AIC                      | $0.9686^{*}$    | $0.9337^{*}$   | $0.9633^{*}$  | $0.9870^{*}$  | 0.9754        | $0.9351^{**}$ | $0.8859^{**}$ | $0.8911^{*}$   | $0.8744^{*}$       | $0.8751^{*}$ |
| direct AIC                        | $0.9686^{*}$    | $0.9473^{**}$  | $0.9889^{**}$ | 1.0713        | 1.2533        | $0.9351^{**}$ | $0.8779^{**}$ | $0.9120^{*}$   | $0.9800^{*}$       | 1.1020       |
| mod. $AIC(NW)$                    | $0.9683^{*}$    | $0.9435^{**}$  | $0.9969^{**}$ | 1.0828        | 1.2589        | $0.9349^{**}$ | $0.8749^{**}$ | $0.9089^{**}$  | $0.9767^{*}$       | 1.1048       |
| mod. AIC(diag)                    | $0.9686^{*}$    | $0.9488^{**}$  | $0.9944^{*}$  | 1.0729        | 1.1980        | $0.9351^{**}$ | $0.8749^{**}$ | $0.9049^{**}$  | $0.9601^{*}$       | 1.0557       |
| iterated BIC                      | $0.9643^{*}$    | $0.9356^{*}$   | $0.9521^{*}$  | $0.9667^{*}$  | 1.0352        | $0.9372^{**}$ | $0.8969^{*}$  | $0.9002^{*}$   | $0.8958^{*}$       | $0.9253^{*}$ |
| direct BIC                        | $0.9643^{*}$    | $0.9429^{**}$  | $0.9658^{**}$ | 1.0519        | 1.2192        | $0.9372^{**}$ | $0.8830^{**}$ | $0.9071^{**}$  | $0.9640^{*}$       | 1.0730       |
| mod. BIC(NW)                      | $0.9600^{**}$   | $0.9489^{*}$   | $0.9864^{*}$  | 1.0725        | 1.2291        | $0.9379^{*}$  | $0.8941^{*}$  | $0.9163^{*}$   | $0.9548^{*}$       | 1.0700       |
| mod. BIC(diag)                    | $0.9643^{*}$    | $0.9491^{**}$  | 0.9875        | 1.1089        | 1.2337        | $0.9372^{**}$ | $0.8843^{**}$ | $0.9027^{*}$   | 0.9666             | 1.0909       |
| The table reports the             | ne MSFE of 1    | the different  | forecasts as  | a ratio of th | ie MSFE of    | the iterated  | AR forecast   | based on th    | ie model sel       | scted by     |
| AIC with $w = 120$ .              | The MSFE i      | s calculated   | only for thos | e periods wl  | nere forecast | ts from all m | lethods are a | vailable. For  | the model          | selection    |
| methods see the foot              | tnote to Tabl   | e 3. * (**) in | dicates that  | the iterated  | AR based o    | m AIC has a   | smaller MSI   | FE in less the | an $50\%$ ( $25$ % | (0) of the   |
| forecasts that are $si_{\hat{b}}$ | gnificantly dif | fferent accord | ling to the G | iacomini-Wl   | nite (2006) t | est.          |               |                |                    |              |

| ) (averages over | •   |
|------------------|-----|
| (A)–(D           |     |
| Category         |     |
| ariables in      | • • |
| e among va       | •   |
| ng performance   | •   |
| forecastir       |     |
| Relative         |     |
| Table 7:         | ļ   |

| Table 8: Relative    | forecastin             | ng perform           | ance amon     | g variabl | es in cat | egory (E    | imoN <sup>"</sup> ( | nal prices     | s, wages, | and    |
|----------------------|------------------------|----------------------|---------------|-----------|-----------|-------------|---------------------|----------------|-----------|--------|
| money" (averages     | over 35 v <sup>ε</sup> | $\tilde{\mathbf{r}}$ |               | )         |           | ,<br>,<br>) |                     | 4              | )         |        |
| Forecast horizon     | 1                      | 3                    | 9             | 12        | 24        | 1           | 3                   | 9              | 12        | 24     |
|                      |                        | A                    | R, w = 120    |           |           |             | A.                  | $R, w = 2^{2}$ | 40        |        |
| iterated AIC         | 1.0000                 | 1.0000               | 1.0000        | 1.0000    | 1.0000    | 0.9935      | 0.9542              | 0.9471         | 0.9890    | 1.1168 |
| direct AIC           | 1.0000                 | $0.9851^{*}$         | $0.9881^{*}$  | 1.0408    | 1.2903    | 0.9935      | 0.9444              | 0.9285         | 1.0063    | 1.2457 |
| mod. AIC(NW)         | 1.0001                 | $0.9849^{*}$         | $0.9899^{*}$  | 1.0461    | 1.3211    | 0.9964      | 0.9471              | 0.9320         | 1.0062    | 1.2441 |
| mod. AIC(diag)       | 1.0000                 | $0.9844^{*}$         | 0.9885        | 1.0516    | 1.3396    | 0.9935      | 0.9437              | 0.9276         | 1.0014    | 1.2692 |
| iterated <b>BIC</b>  | 1.0212                 | 1.1084               | 1.2117        | 1.2787    | 1.2745    | 1.0572      | 1.1938              | 1.3929         | 1.6096    | 1.7898 |
| direct BIC           | 1.0212                 | 1.0485               | 1.0448        | 1.0936    | 1.3066    | 1.0572      | 0.9982              | 0.9727         | 1.0551    | 1.3388 |
| mod. BIC(NW)         | 1.0239                 | 1.0409               | 1.0473        | 1.0883    | 1.3207    | 1.0559      | 1.0301              | 0.9759         | 1.0674    | 1.3342 |
| mod. BIC(diag)       | 1.0212                 | 1.0430               | 1.0516        | 1.1805    | 1.6596    | 1.0572      | 0.9955              | 0.9669         | 1.0626    | 1.6760 |
|                      |                        | Fac                  | tors, $w = 1$ | 20        |           |             | Fact                | ors, $w =$     | 240       |        |
| iterated AIC         | 1.0507                 | 1.1695               | 1.3346        | 1.4491    | 1.4145    | 1.0969      | 1.3154              | 1.5669         | 1.7742    | 1.9435 |
| direct AIC           | 1.0507                 | $1.1385^{*}$         | $1.2858^{**}$ | 1.3614    | 1.3969    | 1.0969      | 1.2109              | 1.3299         | 1.3248    | 1.3001 |
| mod. AIC(NW)         | $1.0437^{*}$           | 1.1339               | $1.2813^{**}$ | 1.3662    | 1.3961    | 1.0935      | 1.2229              | 1.3360         | 1.3377    | 1.2978 |
| mod. AIC(diag)       | 1.0507                 | $1.1325^{*}$         | $1.2461^{**}$ | 1.2753    | 1.4590    | 1.0969      | 1.2119              | 1.3367         | 1.3278    | 1.3342 |
| iterated <b>BIC</b>  | 1.0280                 | 1.1559               | 1.3167        | 1.4406    | 1.4221    | 1.0935      | 1.3407              | 1.6288         | 1.8809    | 2.0442 |
| direct BIC           | 1.0280                 | 1.1303               | 1.2392        | 1.3224    | 1.4204    | 1.0935      | 1.2396              | 1.3591         | 1.3585    | 1.3324 |
| mod. BIC(NW)         | 1.0228                 | 1.1119               | 1.2251        | 1.2968    | 1.4422    | 1.0928      | 1.2489              | 1.3644         | 1.3604    | 1.3558 |
| mod. BIC(diag)       | 1.0280                 | 1.1315               | 1.1911        | 1.3550    | 1.6727    | 1.0935      | 1.2335              | 1.3494         | 1.3826    | 1.6004 |
| See footnote of Tabl | e 7.                   |                      |               |           |           |             |                     |                |           |        |

35

# **CESifo Working Paper Series**

for full list see www.cesifo-group.org/wp (address: Poschingerstr. 5, 81679 Munich, Germany, office@cesifo.de)

- 2481 Thomas Aronsson and Erkki Koskela, Optimal Redistributive Taxation and Provision of Public Input Goods in an Economy with Outsourcing and Unemployment, December 2008
- 2482 Stanley L. Winer, George Tridimas and Walter Hettich, Social Welfare and Coercion in Public Finance, December 2008
- 2483 Bruno S. Frey and Benno Torgler, Politicians: Be Killed or Survive, December 2008
- 2484 Thiess Buettner, Nadine Riedel and Marco Runkel, Strategic Consolidation under Formula Apportionment, December 2008
- 2485 Irani Arraiz, David M. Drukker, Harry H. Kelejian and Ingmar R. Prucha, A Spatial Cliff-Ord-type Model with Heteroskedastic Innovations: Small and Large Sample Results, December 2008
- 2486 Oliver Falck, Michael Fritsch and Stephan Heblich, The Apple doesn't Fall far from the Tree: Location of Start-Ups Relative to Incumbents, December 2008
- 2487 Cary Deck and Harris Schlesinger, Exploring Higher-Order Risk Effects, December 2008
- 2488 Michael Kaganovich and Volker Meier, Social Security Systems, Human Capital, and Growth in a Small Open Economy, December 2008
- 2489 Mikael Elinder, Henrik Jordahl and Panu Poutvaara, Selfish and Prospective: Theory and Evidence of Pocketbook Voting, December 2008
- 2490 Maarten Bosker and Harry Garretsen, Economic Geography and Economic Development in Sub-Saharan Africa, December 2008
- 2491 Urs Fischbacher and Simon Gächter, Social Preferences, Beliefs, and the Dynamics of Free Riding in Public Good Experiments, December 2008
- 2492 Michael Hoel, Bush Meets Hotelling: Effects of Improved Renewable Energy Technology on Greenhouse Gas Emissions, December 2008
- 2493 Christian Bruns and Oliver Himmler, It's the Media, Stupid How Media Activity Shapes Public Spending, December 2008
- 2494 Andreas Knabe and Ronnie Schöb, Minimum Wages and their Alternatives: A Critical Assessment, December 2008
- 2495 Sascha O. Becker, Peter H. Egger, Maximilian von Ehrlich and Robert Fenge, Going NUTS: The Effect of EU Structural Funds on Regional Performance, December 2008

- 2496 Robert Dur, Gift Exchange in the Workplace: Money or Attention?, December 2008
- 2497 Scott Alan Carson, Nineteenth Century Black and White US Statures: The Primary Sources of Vitamin D and their Relationship with Height, December 2008
- 2498 Thomas Crossley and Mario Jametti, Pension Benefit Insurance and Pension Plan Portfolio Choice, December 2008
- 2499 Sebastian Hauptmeier, Ferdinand Mittermaier and Johannes Rincke, Fiscal Competition over Taxes and Public Inputs: Theory and Evidence, December 2008
- 2500 Dirk Niepelt, Debt Maturity without Commitment, December 2008
- 2501 Andrew Clark, Andreas Knabe and Steffen Rätzel, Boon or Bane? Others' Unemployment, Well-being and Job Insecurity, December 2008
- 2502 Lukas Menkhoff, Rafael R. Rebitzky and Michael Schröder, Heterogeneity in Exchange Rate Expectations: Evidence on the Chartist-Fundamentalist Approach, December 2008
- 2503 Salvador Barrios, Harry Huizinga, Luc Laeven and Gaëtan Nicodème, International Taxation and Multinational Firm Location Decisions, December 2008
- 2504 Andreas Irmen, Cross-Country Income Differences and Technology Diffusion in a Competitive World, December 2008
- 2505 Wenan Fei, Claude Fluet and Harris Schlesinger, Uncertain Bequest Needs and Long-Term Insurance Contracts, December 2008
- 2506 Wido Geis, Silke Uebelmesser and Martin Werding, How do Migrants Choose their Destination Country? An Analysis of Institutional Determinants, December 2008
- 2507 Hiroyuki Kasahara and Katsumi Shimotsu, Sequential Estimation of Structural Models with a Fixed Point Constraint, December 2008
- 2508 Barbara Hofmann, Work Incentives? Ex Post Effects of Unemployment Insurance Sanctions – Evidence from West Germany, December 2008
- 2509 Louis Hotte and Stanley L. Winer, The Demands for Environmental Regulation and for Trade in the Presence of Private Mitigation, December 2008
- 2510 Konstantinos Angelopoulos, Jim Malley and Apostolis Philippopoulos, Welfare Implications of Public Education Spending Rules, December 2008
- 2511 Robert Orlowski and Regina T. Riphahn, The East German Wage Structure after Transition, December 2008
- 2512 Michel Beine, Frédéric Docquier and Maurice Schiff, International Migration, Transfers of Norms and Home Country Fertility, December 2008

- 2513 Dirk Schindler and Benjamin Weigert, Educational and Wage Risk: Social Insurance vs. Quality of Education, December 2008
- 2514 Bernd Hayo and Stefan Voigt, The Relevance of Judicial Procedure for Economic Growth, December 2008
- 2515 Bruno S. Frey and Susanne Neckermann, Awards in Economics Towards a New Field of Inquiry, January 2009
- 2516 Gregory Gilpin and Michael Kaganovich, The Quantity and Quality of Teachers: A Dynamic Trade-off, January 2009
- 2517 Sascha O. Becker, Peter H. Egger and Valeria Merlo, How Low Business Tax Rates Attract Multinational Headquarters: Municipality-Level Evidence from Germany, January 2009
- 2518 Geir H. Bjønnes, Steinar Holden, Dagfinn Rime and Haakon O.Aa. Solheim, ,Large' vs. ,Small' Players: A Closer Look at the Dynamics of Speculative Attacks, January 2009
- 2519 Jesus Crespo Cuaresma, Gernot Doppelhofer and Martin Feldkircher, The Determinants of Economic Growth in European Regions, January 2009
- 2520 Salvador Valdés-Prieto, The 2008 Chilean Reform to First-Pillar Pensions, January 2009
- 2521 Geir B. Asheim and Tapan Mitra, Sustainability and Discounted Utilitarianism in Models of Economic Growth, January 2009
- 2522 Etienne Farvaque and Gaël Lagadec, Electoral Control when Policies are for Sale, January 2009
- 2523 Nicholas Barr and Peter Diamond, Reforming Pensions, January 2009
- 2524 Eric A. Hanushek and Ludger Woessmann, Do Better Schools Lead to More Growth? Cognitive Skills, Economic Outcomes, and Causation, January 2009
- 2525 Richard Arnott and Eren Inci, The Stability of Downtown Parking and Traffic Congestion, January 2009
- 2526 John Whalley, Jun Yu and Shunming Zhang, Trade Retaliation in a Monetary-Trade Model, January 2009
- 2527 Mathias Hoffmann and Thomas Nitschka, Securitization of Mortgage Debt, Asset Prices and International Risk Sharing, January 2009
- 2528 Steven Brakman and Harry Garretsen, Trade and Geography: Paul Krugman and the 2008 Nobel Prize in Economics, January 2009
- 2529 Bas Jacobs, Dirk Schindler and Hongyan Yang, Optimal Taxation of Risky Human Capital, January 2009

- 2530 Annette Alstadsæter and Erik Fjærli, Neutral Taxation of Shareholder Income? Corporate Responses to an Announced Dividend Tax, January 2009
- 2531 Bruno S. Frey and Susanne Neckermann, Academics Appreciate Awards A New Aspect of Incentives in Research, January 2009
- 2532 Nannette Lindenberg and Frank Westermann, Common Trends and Common Cycles among Interest Rates of the G7-Countries, January 2009
- 2533 Erkki Koskela and Jan König, The Role of Profit Sharing in a Dual Labour Market with Flexible Outsourcing, January 2009
- 2534 Tomasz Michalak, Jacob Engwerda and Joseph Plasmans, Strategic Interactions between Fiscal and Monetary Authorities in a Multi-Country New-Keynesian Model of a Monetary Union, January 2009
- 2535 Michael Overesch and Johannes Rincke, What Drives Corporate Tax Rates Down? A Reassessment of Globalization, Tax Competition, and Dynamic Adjustment to Shocks, February 2009
- 2536 Xenia Matschke and Anja Schöttner, Antidumping as Strategic Trade Policy Under Asymmetric Information, February 2009
- 2537 John Whalley, Weimin Zhou and Xiaopeng An, Chinese Experience with Global 3G Standard-Setting, February 2009
- 2538 Claus Thustrup Kreiner and Nicolaj Verdelin, Optimal Provision of Public Goods: A Synthesis, February 2009
- 2539 Jerome L. Stein, Application of Stochastic Optimal Control to Financial Market Debt Crises, February 2009
- 2540 Lars P. Feld and Jost H. Heckemeyer, FDI and Taxation: A Meta-Study, February 2009
- 2541 Philipp C. Bauer and Regina T. Riphahn, Age at School Entry and Intergenerational Educational Mobility, February 2009
- 2542 Thomas Eichner and Rüdiger Pethig, Carbon Leakage, the Green Paradox and Perfect Future Markets, February 2009
- 2543 M. Hashem Pesaran, Andreas Pick and Allan Timmermann, Variable Selection and Inference for Multi-period Forecasting Problems, February 2009